

ISSN 2443-8014 (online)

The 2019 Stability & Convergence Programmes

An Overview and Assessment of the Euro Area Fiscal Stance

INSTITUTIONAL PAPER 110 | JULY 2019



European Economy Institutional Papers are important reports analysing the economic situation and economic developments prepared by the European Commission's Directorate-General for Economic and Financial Affairs, which serve to underpin economic policy-making by the European Commission, the Council of the European Union and the European Parliament.

Views expressed in unofficial documents do not necessarily represent the views of the European Commission.

LEGAL NOTICE

Neither the European Commission nor any person acting on behalf of the European Commission is responsible for the use that might be made of the information contained in this publication.

This paper exists in English only and can be downloaded from <u>https://ec.europa.eu/info/publications/economic-and-financial-affairs-publications_en.</u>

Luxembourg: Publications Office of the European Union, 2019

PDF ISBN 978-92-79-98857-8 ISSN 2443-8014 doi:10.2765/939118 KC-BC-19-017-EN-N

© European Union, 2019

Reuse is authorised provided the source is acknowledged. The reuse policy of European Commission documents is regulated by Decision 2011/833/EU (OJ L 330, 14.12.2011, p. 39). For any use or reproduction of material that is not under the EU copyright, permission must be sought directly from the copyright holders.

European Commission Directorate-General for Economic and Financial Affairs

The 2019 Stability & Convergence Programmes

An Overview and Assessment of the Euro Area Fiscal Stance

EUROPEAN ECONOMY

Institutional Paper 110

ACKNOWLEDGEMENTS

This paper was prepared in the Directorate-General of Economic and Financial Affairs under the direction of Marco Buti, Director-General, Carlos Martinez Mongay, Deputy Director-General, and Lucio Pench, Director for Fiscal Policy.

Gilles Mourre was the coordinator of the paper. The main contributors were Alessandra Cepparulo, Stéphanie Pamies, Vito Ernesto Reitano and Klara Stovicek. Other contributors were Pedro Arevalo, Ulrich Clemens, Laszlo Jankovics, Fabrice Orlandi, Adriana Reut and Roberta Torre. Christoph Maier provided valuable help. In addition, the paper benefited from comments and suggestions by Nicolas Carnot, Sven Langedijk, Lucio Pench, Lucia Piana and Eric Ruscher. Comments from DG ECFIN country desks are gratefully acknowledged.

The authors would also like to thank members of the Economic and Financial Committee – Alternates as well as colleagues at the European Central Bank for their insightful comments and corrections.

Comments would be gratefully received and should be sent to:

Gilles Mourre European Commission Directorate-General for Economic and Financial Affairs Directorate for Fiscal Policy Office CHAR 12-040 B-1049 Brussels e-mail: gilles.mourre@ec.europa.eu

Vito Ernesto Reitano European Commission Directorate-General for Economic and Financial Affairs Directorate for Fiscal Policy Office CHAR 12-117 B-1049 Brussels e-mail: <u>vito.reitano@ec.europa.eu</u>

CONTENTS

EXE	CUTIV	/E SUMMARY	7
INTF	RODU	CTION	9
1.	2018	8 AT A GLANCE: BUDGETARY DEVELOPMENTS	11
2.	BUD	GETARY PLANS FOR 2019 AND BEYOND	23
	2.1.	Description of planned budgetary developments	23
	2.2.	Assessment of the euro area fiscal stance	32
		2.2.1. Description of the monetary policy stance and fiscal stance in 2019	32
		2.2.2. What could be an appropriate euro area fiscal stance in 2020?	34
		2.2.3. Assessing the fiscal stance derived from the Stability Programmes	40
3.	RISK	ASSESSMENT OF MEMBER STATES' PLANS	51
	3.1.	Risks to 2019 and 2020 plans	51
	3.2.	Risks to 2021 and 2022 plans	54
4.	SUS	TAINABILITY	57
	4.1.	Short-term fiscal sustainability risks	57
	4.2.	Medium- to long-term fiscal sustainability: government debt projections	59
	4.3.	Medium-term fiscal sustainability challenges	60
	4.4.	long-term fiscal challenges	61
ANI	VEX I.		68

LIST OF TABLES

1.1.	Economic and budgetary developments in the EU and the euro area in 2018	16
1.2.	Expenditure dynamics and medium-term potential GDP growth (% change)	19
3.1.	GDP growth and main fiscal variables: a comparison between 2019 SCPs and Commission	
	forecast	51
3.2.	2019 SCPs - GDP growth and main fiscal variables	56
4.1.	Overall risk classification in the 2019 assessment round, COM no-policy-change scenario	64
A1.1.	Real GDP growth (%)	68
A1.2.	Output gap (% of pot. GDP)	68
A1.3.	General government balance (% of GDP)	69
A1.4.	General government total revenue (% of GDP)	69
A1.5.	General government total expenditure (% of GDP)	70
A1.6.	Structural balance (% of pot. GDP)	70

A1.7. Change in structural balance (pps. of pot. GDP)	71
A1.8. Structural primary balance (% of pot. GDP)	71
A1.9. Change in structural primary balance (pps. of pot. GDP)	72
A1.10.Fiscal effort based on the expenditure benchmark (% of GDP)	72
A1.11.General government total debt (% of GDP)	73

LIST OF GRAPHS

1.1.	Drivers of the change in the headline balance in EU Member States in 2018 (pps. of GDP)	11
1.2.	Fiscal effort based on the expenditure benchmark and change in the structural balance in	
	2018 (pps. of GDP)	12
1.3.	Contributions to the change in the debt-to-GDP ratio in 2018 (pps. of GDP)	14
1.4.	Headline balance in EU Member States in 2018, outturn vs 2018 SCPs (% of GDP)	17
1.5.	Developments of key indicators for the fiscal stance in the euro area (% of pot. GDP), based	
	on Commission 2019 spring forecast	17
1.6.	Discretionary fiscal effort 2017-2019 (% of GDP)	20
1.7.	Expenditure and revenues developments	20
2.1.	Planned budgetary developments according to the 2019 SCPs (% of GDP)	23
2.2.	Composition of the adjustment of the headline balance between 2018 and 2022, as	
	planned in the 2019 SCPs (pps. of GDP)	24
2.3.	The planned change in the (recalculated) structural balance according to the 2019 SCPs	
	(% of potential GDP)	26
2.4.	Progress towards the MTO throughout the programme horizon	26
2.5.	The fiscal effort based on the expenditure benchmark planned in 2019-2021 in the EU	
	Member States according to the 2019 SCPs (%of GDP)	30
2.6.	Contributions to the change in the debt-to-GDP ratio between 2018 and 2022, as planned	
	in the 2019 SCPs (pps. of GDP)The S2 sustainability gap: 'COM no-policy-change' and 'SCP'	
	scenarios (in pps. of GDP)	31
2.7.	Debt level in 2018 vs planned change in structural budget balance in the period 2019-2022	31
2.8.	Monetary policy stance in the euro area	32
2.9.	Fiscal stance and output gap in euro area Member States, 2019	34
2.10.	Fiscal map: Sustainability and stabilisation challenges in 2020 (% of GDP)	35
2.11.	Change in the structural primary balance in 2020: Stability Programmes' plans against	
	sustainability and stabilisation needs (% of GDP)	36
2.12.	Fiscal stance scenarios in 2020 based on the change in the structural balance	38
2.13.	Fiscal stance scenarios in 2020 based on the % change in net primary expenditure	39
2.14.	Euro area real GDP forecast - Uncertainty linked to the balance of risks	40
4.1.	Short-term fiscal sustainability indicator (S0 indicator) and sub-indexes	59
4.2.	Medium-term government debt projections for the EU-28 (in % of GDP)	60
4.3.	The \$1 indicator (fiscal adjustment required to reach a 60% public debt to GDP ratio by	
	2033, in pps. of GDP)	61

4.4.	The S2 sustainability gap decomposed (in pps. of GDP)	62
4.5.	The S2 sustainability gap: 'COM no-policy-change' and 'SCP' scenarios (in pps. of GDP)	63

LIST OF BOXES

1.1.	Debt and deficit developments in the euro area: comparison with the US and JP	14
1.2.	Euro area fiscal stance according to IMF, OECD and European Commission	21
2.1.	Comparison with the Draft Budgetary Plans of euro area Member State	25
2.2.	Progress towards the MTO	27
2.3.	Update of the minimum MTO	29
2.4.	Stabilisation needs in the euro area as identified by a double condition on unemployment	41
2.5.	Alternative measure of fiscal sustainability challenges for the fiscal stance analysis	44
2.6.	Are SCPs moving toward growth-friendlier composition of expenditure and revenue? A	
	tentative score-based assessment	46
2.7.	Public investment in the euro area	48
3.1.	Independent production or endorsement of the macroeconomic forecasts underpinning	
	the 2019 Stability and Convergence Programmes	52
4.1.	The Commission approach to assess fiscal sustainability	58
4.2.	Fiscal sustainability under market-based interest rate expectations assumptions	65

EXECUTIVE SUMMARY

This paper provides a review of the Stability and Convergence Programmes (SCPs) that the Member States of the European Union (EU) submitted in spring 2019. After examining budgetary developments in 2018, the paper analyses fiscal projections for 2019 and medium-term plan for the 2020-2022 period. It also presents the implications for budgetary developments at euro area and EU aggregate levels and a detailed assessment of the euro area fiscal stance.

Fiscal outcomes in 2018

The budgetary situation continued to improve in 2018, bringing the aggregate deficit to its lowest level since 2000. In aggregate terms, the deficit declined for the eighth consecutive year, to 0.6% in the EU, with around half of Member States achieving a budget surplus.

The deficit reduction took place despite some moderate fiscal loosening and was attributable to favourable cyclical conditions, positive revenue developments and lower interest expenditures. At aggregate level, the budget deficit declined by 0.4 pps. in the EU. Real GDP growth above potential growth contributed to this decline, raising the cyclical component of the budget. The structural balance (i.e. the balance corrected for the business cycle and one-offs) also improved, thanks to favourable revenue developments and lower interest expenditure. By contrast, the expenditure benchmark, which is not affected by revenue windfalls and interest expenditure, points to a negative fiscal effort in 2018. Revenue windfalls also explain the larger-than-planned deficit reduction at aggregate level compared with the objectives in the 2018 SCPs, despite real GDP growing less than expected. Fiscal performances remained mixed at country level in 2018: 12 Member States recorded a structural balance above their medium-term budgetary objective (MTO), whereas the distance to the MTO widened in 9 Member States.

Fiscal projections for 2019 and 2020-22

The budgetary objectives in the SCPs imply an increase in the aggregate deficit in 2019, while the fiscal adjustment planned for the following years would result in an aggregate deficit in 2022 slightly below the 2018 level. Based on the SCP projections, the deficit would rise to 0.9% of GDP in both the EU and the euro area in 2019. It is projected to decline afterwards, reaching 0.4% of GDP in the EU and 0.3% in the euro area in 2022. However, at aggregate level, the expenditure benchmark point to a fiscal effort close to zero between 2019 and 2022.

In 2019, the SCPs project a negative fiscal effort based on the expenditure benchmark as well as a worsening in the structural balance at aggregate level. Based on the SCPs, the expenditure benchmark points to a negative fiscal effort of more than ¼ pp. of GDP this year. The worsening in the structural balance would be slightly smaller. In most countries below the MTO in 2018, the planned fiscal adjustment is backloaded, as the structural balance is not projected to improve in 2019.

The planned structural adjustment would not be sufficient to achieve the MTO in some highlyindebted Member States. While according to the SCPs the number of Member States at or above the MTO would increase from 12 in 2018 to 23 in 2022, some highly-indebted countries such as Spain, France and Italy plan to remain far from their MTO at the end of the programme period. As for the 12 Member States that have already reached their MTO in 2018, 9 plan a fiscal expansion over the programme horizon, but they would all remain at or above their MTO in 2022.

Risks to the SCP budgetary projections

While downside risks to the SCP budgetary projections for 2019 appear to be small based on the Commission 2019 spring forecast, sizeable consolidation measures are needed in some Member States to reach the 2020 budgetary targets. For 2019, the budgetary projections in Commission forecast are in line with the SCPs for the euro area and only marginally more negative for the EU. Also at country level, only four Member States have submitted budgetary projections for 2019 significantly more favourable than the Commission forecast, mainly due to different assumptions regarding the impact of

discretionary measures. Based on the no-policy-change Commission forecast, the gap to reach the planned targets in 2020 is close to ½ pp. of GDP at aggregate level and close or above 1 pp. of GDP in six Member States. Hence, the SCP budgetary targets for 2020 seem to be subject to sizeable downside risks.

The risks to the fiscal adjustments planned beyond 2020 appear contained. First, the revenue projections seem to be rather prudent, with some projected revenue shortfalls at EU level. Second, no further savings from lower interest expenditure are projected at aggregate level. Third, the required measures needed to meet the SCP objectives are reasonable in size on aggregate. However, as stated above, the SCPs targets are not particularly ambitious in some Member States with high debt levels.

Implications of debt developments and sustainability risks

In 2018, the aggregate debt-to-GDP ratio fell by around 2 pps, to 81.5% in the EU and 87.1% in the euro area. At aggregate level, the decline in the debt ratio was driven by a primary surplus above 1% of GDP and nominal GDP growing more than the implicit interest rate paid on debt, implying a debt-decreasing snowball effect. The euro area and EU aggregate debt ratios have steadily fallen since the peak of over 94% and 88% of GDP reached in 2014. By comparison, in 2018 the public debt was around 107% of GDP in the US and 236% of GDP in Japan.

The SCPs project a further decline in the debt-to-GDP ratios in most Member States over the programme horizon. Those projections imply a reduction in the aggregate debt ratio by around 7 pps. of GDP by 2022, to around 75% in the EU and 80% in the euro area. The decline of the debt ratios continues to be driven by primary surpluses and debt-decreasing snowball effect. At country level, the snowball effect would be debt-increasing only in Italy. France and Italy plan to reduce the gap to the 60% threshold by less than 1/20th per year on average over the programme period.

Medium-term fiscal sustainability risks would remain high in five Member States after the implementation of the SCP plans. At aggregate EU level, the improvement in the structural primary balance required to achieve a debt-to-GDP ratio of 60% by 2033 relative to the Commission no-policy-change forecast for 2020 amounts to 1.9 pps. of GDP over the period 2021–2025. If fiscal plans in the SCPs were fully implemented, the fiscal gap would be reduced to 0.8 pps. However, Belgium, Spain, France, Italy and Portugal would remain at high risk for medium-term fiscal sustainability.

The fiscal stance for the euro area: direction and appropriateness

After being broadly neutral in 2015-2017, the euro area fiscal stance turned slightly expansionary in in 2018, based on the discretionary fiscal effort. The discretionary fiscal effort, an indicator conceptually close to the expenditure benchmark of the Stability and Growth Pact, signals slightly expansionary discretionary fiscal policies in the euro area in 2018. By contrast, the change in the structural balance indicates a slightly contractionary fiscal stance; however, such change benefited from large revenue windfalls and lower interest expenditure.

The Commission 2019 spring forecast points to an expansionary euro area fiscal stance in 2019 and, under unchanged policies, in 2020. The discretionary fiscal effort indicates an expansionary fiscal stance in 2019-2020, as net primary spending is projected to increase more than medium-term potential growth, also due to discretionary tax cuts. The change in the structural balance points to a slightly expansionary fiscal stance in both years.

An appropriate differentiation of national fiscal policies consistent with the proposed Country-Specific Recommendations would lead to a broadly neutral fiscal stance for the euro area in 2020. The combination of restrictive fiscal policies, in line with the Country-Specific Recommendations (CSRs), in Member States that are required to adjust and expansionary fiscal policies in Member States that are expected to overachieve their MTO in 2019 would lead to a broadly neutral euro area fiscal stance in 2020.

INTRODUCTION

This paper provides an overview of the 2019 Stability and Convergence Programmes (SCPs) submitted by EU Member States. (¹) The paper aims at offering a cross-country aggregated view of fiscal policy plans in the European Union and the euro area as a whole. It also includes an assessment of the fiscal stance and policy mix in the euro area.

In its 2019 Annual Growth Survey the Commission highlighted that credible actions to achieve agreed fiscal objectives, in line with the common European rules, remain essential. A number of Member States have reduced their public debt and achieved or exceeded their medium-term budgetary objective, creating scope for higher public investment to support potential growth. However, several others continue to shoulder high levels of public debt, which constrain their ability to invest for the future. These countries have also made less progress in reducing public debt over recent years. The (current) economic expansion in those countries should be used to build up buffers, further strengthen their public finances, in particular in structural terms; and prioritise expenditure on items that foster resilience and growth potential. The Annual Growth Survey also stressed the importance of improving the quality and composition of public finances and ensuring long-term sustainability of public finances.

The 2019 Euro Area Recommendation on fiscal policies adopted by the Commission and the Council called for a differentiation of fiscal policies depending on fiscal space. On 22 January 2019, the Council recommended that in the period 2019-2020, euro area Member States, while pursuing policies in full respect of the Stability and Growth Pact, support public and private investment and improve the quality and composition of public finances. It is further recommended to rebuild fiscal buffers, especially in euro area countries with high levels of public debt. This horizontal recommendation feeds into the Country-Specific Recommendations. In the context of the European Semester, the Council recommendations, both horizontal and country-specific, are expected to guide the national budgets for 2020. For this reason, plans for 2020 are given primary attention in the present note.

This note consists of four sections. Section 1 examines the implementation of SCPs in 2018. Section 2 presents the budgetary plans set out by Member States in their SCPs over the period 2019 to 2022. It also analyses and assesses the overall fiscal stance in the euro area. Section 3 contains an analysis of the risks to the budgetary trajectories contained in the SCPs. For 2019-2020 it focuses on risks to SCP budgetary targets relative to the Commission 2019 spring forecast, while for 2021-2022 revenue targets as well as interest rate risks are assessed. Section 4 looks at the longer-term implications of the plans for fiscal sustainability, by taking into account the projected changes in age-related expenditure. Finally, an annex provides tables with data from both the SCPs and the Commission 2019 spring forecast.

⁽¹⁾ The analysis is built around data reported by Member States in their 2019 SCPs, unless otherwise specified.

1. 2018 AT A GLANCE: BUDGETARY DEVELOPMENTS

Overall budgetary developments in 2018

In aggregated terms, public deficits decreased for the eighth year in a row and reached the lowest level since 2000. The aggregate headline deficit fell from 1.0% of GDP in 2017 to 0.6% in 2018 in the EU, and from 1.0% to 0.5% in the euro area. As shown in Graph 1.1, the deficit reduction was driven by the improvement in cyclical conditions (i.e. actual real GDP growth above medium-term potential growth), which had a deficit-decreasing impact of around 0.4 pps. of GDP in both the EU and the euro area. Lower interest expenditure (0.1 pp. of GDP) and other factors (0.3 pps. of GDP, essentially revenue windfalls) also supported the deficit reduction in both the EU and the euro area in 2018. Conversely, the fiscal effort (based on the expenditure benchmark) had a deficit increasing impact of around 0.2 pps. of GDP in the EU and 0.3 pps. in the euro area. (²)



Note: A positive (negative) value means contributing to an improvement (worsening) of the headline balance. The 'other' category is defined as a residual and includes: (i) revenue windfalls (+) or shortfalls (-), i.e. revenue developments not explained by nominal GDP growth or discretionary revenue measures; (ii) the impact of smoothing Gross Fixed Capital Formation (GFCF) in the expenditure benchmark; and (iii) a possible different impact of cyclical unemployment benefits used in the expenditure benchmark and in the cyclical component. Here, the change in the cyclical component is based on the medium-term potential growth used to compute the expenditure benchmark. Cyprus falls outside of the graph as the headline balance worsened by 6.5 pps. of GDP in 2018, due to a large one-off support to the banking sector (in the order of 8% of GDP). Data for the UK refer to the financial year 2018/2019. **Source:** European Commission 2019 spring forecast

Headline balances improved in about two thirds of Member States in 2018 compared to 2017, primarily due to favourable cyclical conditions and revenue windfalls (Graph 1.1). However, the headline balance fell by at least 0.3 pps. of GDP in MT, DK, CZ, HR, SE, RO and LV, primarily due to a sizeable negative fiscal effort (based on the expenditure benchmark), and by 6.5 pps. of GDP in CY, due to a large one-off impact of bank recapitalisation. All Member States (except DE and DK) benefited from improving cyclical conditions, as in 2018 actual real GDP growth turned out above (medium-term) potential growth. In contrast, the fiscal effort based on the expenditure benchmark gave a negative

^{(&}lt;sup>2</sup>) For additional information on the computation of the expenditure benchmark see European Commission (2019). "Vade Mecum on the SGP". *European Economy, Institutional Paper 101*, pp 28-32. Available online: <u>https://ec.europa.eu/info/sites/info/files/economy-finance/ip101_en.pdf</u>

contribution to the change in the headline balance in most Member States. Only in CY, BG, IE, FR, HU, PL and UK the fiscal effort was positive. An increase in interest expenditure was recorded only in EL.

The expenditure benchmark and the structural balance provide contrasting signals about the aggregate fiscal effort in 2018 (Graph 1.2). The aggregate structural balance improved in 2018, by 0.2 pps. of GDP in the EU and by 0.3 pps. in the euro area. Conversely, as stated above, the fiscal effort (based on the expenditure benchmark) is estimated to have worsened by 0.2 pps. of GDP in the EU and by 0.3 pps. in the euro area. The difference between the two indicators of fiscal effort is partly explained by lower interest expenditure, which is not included in the expenditure benchmark. In addition, the 2018 structural balance benefited from revenue windfalls and a higher potential growth. A detailed discussion on the euro area fiscal stance in 2018 can be found at the end of this section.



Note: The fiscal effort based on the expenditure benchmark is computed here by using outturn data for GDP deflators in 2018, while the Country-Specific Recommendations for expenditure benchmark developments in 2018 were based on GDP deflators as projected in the Commission 2017 spring forecast. Data for the UK refer to the financial year 2018/2019. **Source:** European Commission 2019 spring forecast

The size and direction of the fiscal effort in 2018 differs considerably across Member States and according to the chosen indicator (Graph 1.2). The change in the fiscal effort based on the expenditure benchmark varied between -2.1 pps. (MT) and +2.0 pps. of GDP (CY), while the change in the structural balance varied between -1.8 pps. (MT) and +0.9 pps. of GDP (PT). In 6 out of the 11 Member States that posted an increase in the structural balance, the fiscal effort based on the expenditure benchmark pointed to the opposite direction (DE, BE, AT, NL, LU and PT). Differences between the two indicators larger than 0.5 pps. of GDP were recorded in around half of the Member States. This divergence is partly

explained by large (at least 0.5 pps. of GDP) revenue windfalls (CZ, DE, ES, LU, PT, PL and RO) or shortfalls (DK, EL, FI and HU). (³)

12 Member States recorded a structural balance above the MTO in 2018, but 9 further departed from it. Among the 11 Member States that based on the Commission 2019 spring forecast were above the MTO already in 2017, 5 (CZ, DK, HR, MT and SE) maintained a structural position above the MTO despite a worsening structural balance, while 6 either further increased or recorded no change in their structural balance (BG, DE, CY, LT, LU and NL). In addition, AT increased the structural balance and achieved its MTO in 2018. EL, which after the economic adjustment programme has nominated an MTO of +0.25% of GDP as of 2020, recorded a structural surplus of 5% of GDP in 2018. As for the other 14 Member States in the preventive arm, 3 (PL, PT and UK) made some progress towards their MTO in 2018, 2 (BE and FR) recorded a similar structural balance than in 2017, while 9 (EE, IE, IT, LV, HU, RO, SI, SK and FI) worsened their structural position. Similar indications come from the fiscal effort based on expenditure benchmark for these 14 Member States: a negative effort was recorded in 9 countries (BE, EE, IT, LV, PT, RO, SI, SK and FI) and a positive effort in 5 (IE, FR, HU, PL and UK). Finally, ES corrected its excessive deficit in 2018 but with no fiscal effort based on the structural balance and a negative effort based on the expenditure benchmark.

Debt-to-GDP ratios continued to decrease in 2018, standing at around 82% and 87% in the EU and the euro area respectively. (4) Public debt peaked in 2014 at around 88% of GDP in the EU and 94% in the euro area, and declined steadily over 2015-2017. In 2018, it decreased by 2.0 pps. of GDP in the EU and 1.7 pps. in the euro area (to 81.5% and 87.1% of GDP, respectively). The decline was driven, similar as in 2017, by primary surpluses and debt-decreasing snowball effect, the latter due to GDP growth exceeding the implicit average cost of servicing the outstanding debt (i.e. r<g) (Graph 1.3). In recent years, public debt developments have been less favourable in the US and Japan (Box 1.1). Most Member States recorded a primary surplus in 2018, which contributed to debt reduction. The exceptions are FR, ES, LV, RO and EE, which recorded primary deficits also in 2017, and CY and LV. The snowball effect turned debt-reducing in all Member States but IT and, to a lesser extent, DK. Stock-flow adjustments, which mainly refer to financial transactions and discrepancies between cash and ESA2010 figures, were broadly neutral in aggregate terms in the EU and the euro area. However, they had a substantial upward effect on public debt of more than 1% of GDP in LU, MT, HU, UK, SI, IE and PL, and especially in CY where the debt-increasing impact was 7% of GDP. By contrast, the stock-flow adjustment had a sizeable downward impact in LT, LV and AT. All in all, in 2018 debt ratios fell in all Member States except CY and IT (where it increased) and in FR and the UK (no change).

^{(&}lt;sup>3</sup>) Compared to revenue developments that would have been expected based on an elasticity of one relative to nominal GDP growth, excluding discretionary revenue measures and revenues from EU funds as estimated in the Commission 2019 spring forecast.

^{(&}lt;sup>4</sup>) Aggregate debt figures are not consolidated for intergovernmental lending amounting to around 1.5% of GDP in the EU and 2% of GDP for the euro area. Excluding intergovernmental lending, in 2018 the debt ratio was 80.0% of GDP in the EU (81.7% in the 2017) and 85.1% in the euro area (87.1% in 2017).



Note: The graph disaggregates the changes in debt-to-GDP ratios in 2018 between the contributions of the primary balance, stock-flow adjustments and the snowball effect, the latter of which refers to the interest rate-growth rate differential. Negative (positive) values indicate that the concerned factor contributed to a decrease (increase) in the debt-to-GDP ratio, i.e. primary balances are shown with an opposite sign. Data for the UK refer to the financial year 2018/2019. **Source:** European Commission 2019 spring forecast

Comparison between the 2018 outturn and the 2018 Stability and Convergence Programmes

Compared to the 2018 Stability and Convergence Programmes (SCPs), the aggregate headline balance turned out better than planned in 2018, despite lower economic growth (Table 1.1). Compared to the 2018 SCPs, real GDP turned out worse than expected in both the EU and the euro area. As a result, the output gap increased by less than projected in the SCPs (the level of the output gap turned out higher due to its upward revision in 2017). At the same time, the GDP deflator turned out broadly as expected. Nevertheless, the headline balance improved by more than projected in the 2018 SCPs mainly due to higher-than-planned revenues. Similar, structural balance turned out better than projected, driven by revenue windfalls. In contrast, the fiscal effort (based on the expenditure benchmark) turned out as projected in both the EU and the euro area.

Box 1.1: Debt and deficit developments in the euro area: comparison with the US and JP

On aggregate, deficit levels are consistently smaller in the euro area than in the US and Japan (Graph 1). Over the period 2013-2018, euro-area deficit levels were on a declining trend while economic conditions were improving. On average, in this period the euro area had somewhat larger real GDP growth than Japan (1.6% of GDP and 1.2% of GDP, respectively) while posting considerably smaller deficit (-1.8% of GDP and -4.3% of GDP, respectively). Over the same period, the US had the highest real GDP growth on average (2.3%), yet also the largest deficit (-5.2% of GDP). These differences in budgetary developments are expected to continue in the forecast period.



Graph 1: Government balance and macroeconomic developments for EA, US and JP

Debt levels in the euro area are on a declining trend, remaining well below the debt levels in Japan and the US. The debt level in the euro area declined from 94% of GDP in 2013 to 87% of GDP in 2018. Over the same period, debt-to-GDP ratio slightly increased in the US and Japan, reaching 107% and 236% of GDP respectively in 2018. Until 2008, the US had smaller debt-to-GDP ratio than the euro area (Graph 2a). In 2008, the negative gap closed and turned positive. The positive gap kept increasing, reaching more than 20 pps. of GDP in 2018. Over the same period, the positive gap between the euro area and Japan increased even more, from 107 pps. of GDP in 2005 to almost 150 pps. of GDP in 2018. The widening gaps between the euro area and the US/Japan stem from a smaller increase in debt in the euro area during the crisis period (see Graph 2b). In addition, since 2014, the debt to GDP ratio is on a declining trend in the euro area, driven by the decline in Germany and the Netherlands, while it moderately increased in Japan and the US.



Graph 2: Debt developments in the EA, the US and JP

Source: European Commission 2019 spring forecast

The higher-than-projected fall in the 2018 headline deficit of 0.3 pps. of GDP in the EU and 0.2 pps. in the euro area stems from better-than-planned revenue developments. Revenue windfalls largely explain a better-than-projected change in the structural balance in 2018 (by around $\frac{1}{2}$ pp. of GDP), while interest expenditure was broody similar as projected in the SCPs. By contrast, cyclical conditions turned out less buoyant than expected, implying a lower positive impact on the budget balance (of around 0.2 pps. of GDP).

		2018 SCP planned	COM spring forecast
EU	Deal CDD arouth	2.4	2.0
Euro area	Real GDP growin	2.4	1.9
EU	Nominal CDP growth	4.0	3.6
Euro area	Nominal ODP grown	3.9	3.3
EU	Inflation	1.5	1.6
Euro area	(GDP deflator)	1.5	1.4
EU	Output gap	0.5	0.7
Euro area	(% pot. GDP)	0.4	0.6
EU	Headline balance	-0.9	-0.6
Euro area	(% GDP)	-0.7	-0.5
EU	Interest expenditure	1.9	2.0
Euro area	(% GDP)	1.8	1.8
EU	Change in output gap	0.6	0.3
Euro area	(pps. pot. GDP)	0.8	0.4
EU	Change in headline	0.1	0.4
Euro area	balance (% GDP)	0.2	0.3
EU	Change in SB	-0.1	0.3
Euro area	(pps. pot. GDP)	-0.2	0.3
EU	Fiscal effort based on	-0.2	-0.2
Euro area	EB (% GDP)	-0.3	-0.3

Note: Aggregate for the EU and the euro area do not include figures for EL as in 2018 the country did not submit the SP. **Source:** European Commission 2019 spring forecast and 2018 SCPs

Two thirds of Member States reached or over-performed the 2018 headline balance target set in their 2018 SCPs (Graph 1.4). The target was over-performed by at least 0.5 pps. of GDP in nine Member States, as a result of higher than planned structural improvement (BG, PL, LU, MT, DE, NL, HR and AT) and higher than planned one-offs (DK), despite the less positive impact of the cycle on the budget outcome than assumed in last year's SCPs (DK, HR, DE and LU). In contrast, the target was underperformed by at least 0.5 pps. of GDP in four Member States (CZ, EE, IT and CY), primarily because of larger-than-planned structural deterioration (CZ and EE) or due to higher interest expenditure and lower growth (IT). In CY, one-off support measures to the banking sector in the order of 8% of GDP explain why the headline target for 2018 has not been reached.



Notes: The graph plots the notified 2018 headline budget balances (vertical axis) against the planned headline budget balance (horizontal axis). Member States above (below) the 45-degree line are those where the 2018 outcome was better (worse) than planned. CY falls outside the scope of the graph as it notified a headline deficit of 4.8% of GDP compared to a planned surplus of 1.7% of GDP due to large one-offs.

Source: European Commission based on Commission 2019 spring forecast and 2018 SCPs

Developments of the euro area fiscal stance in recent years

Several indicators are used in literature to assess the fiscal stance. $(^5)$ In this paper, three fiscal stance measures are considered: the discretionary fiscal effort, $(^6)$ the change in the structural balance and the change in the structural primary balance. $(^7)$

The euro area fiscal stance was mildly expansionary in 2018, based on the discretionary fiscal effort (DFE). The DFE, conceptually close to the expenditure benchmark that has become the operational indicator of compliance in the preventive arm of the SGP, shows that fiscal policies supported economic activity in 2018 (Graph 1.5). This followed a contractionary and pro-cyclical fiscal stance in 2011-2014, and an overall broadly neutral fiscal stance in 2015-2017. (⁸) The change in the fiscal stance has helped the economic recovery in the euro area since 2015, in a period in which monetary policy has been

^{(&}lt;sup>5</sup>) See for a review IMF (1995). "Guidelines for Fiscal Adjustment". *Pamphlet Series No. 49*; European Commission (2016). "Report on Public Finances in EMU". *European Economy, Institutional Paper 045*, chapter IV.

^{(&}lt;sup>6</sup>) The DFE is based on the budgetary developments that are deemed to be under the control of governments. On the expenditure side, the DFE looks at the increase in primary expenditure – net of one-offs and cyclical expenditures – relative to medium-term potential growth. On the revenue side, it considers discretionary revenue measures – net of one-offs – as assessed in the Commission forecast and in the Stability Programmes. See for more details Carnot, N. and de Castro, F. (2015). "The Discretionary Fiscal Effort: an Assessment of Fiscal Policy and its Output Effect". *European Commission, Economic Papers 543.*

^{(&}lt;sup>7</sup>) The change in the structural (primary) balance, although capturing the broad economic effects of fiscal policy, can be distorted by the following flaws: (i) swings in fiscal elasticities (i.e. tax windfalls or shortfalls); (ii) revisions in the estimations due to difficulties in real time measurement of the output gap; and (iii) effects outside the control of governments (e.g. change in interest rates affecting the structural balance).

^{(&}lt;sup>8</sup>) Given the uncertainty around the measure of structural balance as an unobserved variable and to avoid risks of fine-tuning, an interval of the fiscal stance between -0.2% and 0.2% is considered to be broadly neutral.





Note: The measures of the fiscal stance presented here are: the discretionary fiscal effort (DFE), the change in structural balance (SB) and the change in structural primary balance (SPB). **Source:** European Commission 2019 spring forecast

constrained at the zero lower bound. Looking more in detail at the budgetary composition of the DFE in 2018 (Graph 1.6), primary expenditure – net of one-offs and cyclical unemployment benefits – provided a fiscal impulse to the euro area economy of around 0.3 pps. of GDP. This is the result of primary expenditure increasing more than medium-term nominal potential growth in 2018 (3.3% as against 2.5%) (Table 1.2) $\binom{9}{10}$ These developments follow no fiscal impulse from primary expenditure in 2017. On the revenue side, $\binom{11}{11}$ the fiscal impulse form discretionary revenue measures was nil in 2018, as in 2017. Box 1.2 presents a comparison with the euro area fiscal stance based on IMF and OECD data.

At country level, the DFE points to a broadly neutral fiscal stance in Germany and France in 2018, while Spain and Italy ran expansionary fiscal policies (Graph 1.6). Digging into the budgetary composition of the DFE in large euro area countries, Germany and France show an increase in primary expenditure – net of one-offs and cyclical unemployment benefits – consistent with medium-term potential growth (see Table 1.2), entailing no fiscal impulse from expenditure policies in 2018, as in 2017. By contrast, in 2018 primary expenditure increased significantly above medium-term potential

^{(&}lt;sup>9</sup>) It can be noted that the 'benchmark' nominal growth rates used in this framework reflect the weakness of the 10-year average potential growth estimates, which still incorporate lagged effects from the crisis, as well as the low inflation environment. In 2018, the lower potential growth used in the DFE explains part (around 0.15 pps. of GDP) of the more contractionary fiscal stance of this indicator compared with the fiscal stance measured through the change in the structural balance.

^{(&}lt;sup>10</sup>) It should be noted that the headline expenditure-to-GDP ratio continued to decline in the euro area in 2018 (Graph 1.7). However, this was due to decreasing interest expenditure and favourable cyclical conditions, as GDP growth outpaced potential growth. The structural expenditure ratio has been broadly stable since 2016.

^{(&}lt;sup>11</sup>) On the revenue side, the DFE only takes into account the change in revenues related to discretionary measures (excluding oneoffs), and thus it is not affected by revenue windfalls/shortfalls.

growth in Italy and in Span, implying loose expenditure policies. (¹²) On the revenue side, discretionary tax policies were slightly contractionary in Italy, only partly compensating the expansionary policies on the expenditure side, and broadly neutral in Germany, Spain and France.

	EA		DE			ES			FR			IT			
	2017	2018	2019	2017	2018	2019	2017	2018	2019	2017	2018	2019	2017	2018	2019
Primary expenditure growth (nominal)	2.3	3.3	3.4	3.5	3.7	5.2	2.5	4.2	4.1	2.1	1.9	0.5	1.0	2.7	2.1
Medium term potential growth (nominal) of which:	2.1	2.5	2.8	3.0	3.4	3.7	1.8	1.6	2.3	1.7	2.0	2.4	0.5	0.9	0.8
Medium term potential growth (real):	1.0	1.1	1.2	1.5	1.5	1.6	0.5	0.6	0.7	1.1	1.1	1.1	0.0	0.0	0.2
GDP deflator	1.1	1.4	1.6	1.5	1.9	2.1	1.2	1.0	1.6	0.7	0.9	1.3	0.5	0.8	0.7
ote: Primary expenditure is net of one-offs and cyclical unemployment expenditure															

 Table 1.2:
 Expenditure dynamics and medium-term potential GDP growth (% change)

Note: Primary expenditure is net of one-offs and cyclical unemployment expenditu Source: European Commission 2019 spring forecast

The change in the structural balance, indicating a mildly contractionary euro area fiscal stance in **2018**, appears to be distorted by interest savings and large revenue windfalls. Compared with the DFE, the 2018 structural balance benefited from lower interest expenditure, revenue windfalls, and a higher potential growth. More specifically, the ¹/₄ pp. of GDP increase in the structural balance in 2018 was driven by a decline in interest expenditure of 0.1 pp. of GDP and a higher structural revenue ratio of 0.3 pps. of GDP (Graph 1.7). The increase in the latter is largely explained by revenue windfalls. (¹³) Part of these windfalls were however spent as shown by the 0.1 pp. of GDP increase in the euro area structural expenditure ratio. Looking at the geographical composition of the euro area fiscal stance, based on the change in the structural balance the fiscal stance appears to have been contractionary in Germany in 2018, largely due to significant revenue windfalls (0.7% of GDP) originated from a tax rich composition of economic growth. Spain also benefited from sizeable revenue windfalls in 2018 (0.7% of GDP), but the country did not improve its structural balance as those windfalls were spent. The fiscal stance was broadly neutral also in France, Italy and the Netherlands, based on the change in the structural balance.

^{(&}lt;sup>12</sup>) According to the DFE, in 2018 the fiscal expansion due to primary expenditure increasing above medium-term potential growth was around ³/₄ pp. in Italy and 1 pp. in Spain. However, it should be noted that for Italy and Spain the 10-year medium-term potential growth used in the DFE for 2018 (0.0% and 0.6% respectively) is significantly lower than the point estimate of real potential growth for that year (0.5% and 1.2% respectively).

^{(&}lt;sup>13</sup>) In 2018, in the euro area revenues - net of discretionary measures - increased more than nominal GDP, pointing to revenue windfalls of around 0.3% of GDP.



Note: The impact on the DFE of primary expenditure and discretionary revenues is net of one-offs. Primary expenditure is also netted of cyclical unemployment benefits. In 2019, data for France are affected by the reform of CICE (crédit d'impôt pour la compétitivité et l'emploi). As of 2019, CICE has been transformed into a permanent decrease in social contribution, implying lower discretionary revenues. At the same time, subsidies to firms have been reduced, implying lower permanent primary expenditure. Excluding the CICE reform, the impact of primary expenditure on the DFE of France would be reduced to 0.1 pp of GDP (from 1 pp.), while discretionary revenues would provide a contribution of -0.3 pps. of GDP (from -1.2 pps.) **Source:** European Commission 2019 spring forecast



Box 1.2: Euro area fiscal stance according to IMF, OECD and European Commission

Using the change in the structural primary balance $(^1)$ as a measure to assess the euro area fiscal stance, developments over the horizon 2011-2019 are presented in this Box according to the IMF, OECD $(^2)$ and Commission most recent forecasts. A consensus is found on the sign $(^3)$ of the fiscal policy stance all over the period as well as in gauging the sign of the output gap up to 2017 (Graph 1 a and c). Minor discrepancies are observed when looking at the size of the fiscal stance as well as the change in the output gap. In terms of slacks, instead, on average, OECD estimates appear more optimistic about the level of potential GDP (+0.6 pps. on average over the period) while minor differences emerge when considering IMF data (-0.2 pps. on average over the period).



Source: European Commission on OECD Economic outlook May 2019, IMF World economic outlook April 2019 and Commission 2019 spring forecast

After the contractionary and pro-cyclical fiscal policy experienced in 2011-2013, according to the IMF and the OECD the fiscal stance turned broadly neutral in the following five years (2014-2018). The largest differences compared with the fiscal stance based on Commission data are in 2012, when the IMF and OECD estimate a more contractionary stance, and in 2016, when the Commission fiscal stance is more expansionary. Broad agreement is found again for 2019 when all the three forecasters project a slightly expansionary stance (-0.31 pps. of GDP for the IMF; -0.33 pps. for the European Commission and -0.36 pps. for the OECD).

When looking at the average differences at country level in the period 2011-2019, the largest divergences between the Commission and the IMF and the OECD relate to ES and, to a lesser extent, NL (Graph 2a).

(Continued on the next page)

^{(&}lt;sup>1</sup>) For comparability reasons the measure for fiscal stance used by the IMF and the OECD is presented here.

^{(&}lt;sup>2</sup>) Because of missing data for 5 countries (MT, CY, LV, LT and SK), OECD euro area refers to 14 countries.

^{(&}lt;sup>3</sup>) Except in 2018; however, in that year, the fiscal stance remains in the broadly neutral range (-0.2 pps. / + 0.2 pps. of GDP) for the three forecasters.



2. BUDGETARY PLANS FOR 2019 AND BEYOND

2.1. DESCRIPTION OF PLANNED BUDGETARY DEVELOPMENTS

The aggregate budget balance is planned to increase in the EU and the euro area by 0.3 pps. of GDP between 2018 and 2022 to reach a deficit of 0.3% of GDP in 2022 (Graph 2.1). By contrast, starting from a lower budget balance than actually achieved in 2018 (see Table 1.1), last year's Stability and Convergence Programmes (SCPs) planned a surplus of 0.2% of GDP in the EU and 0.5% in the euro area to be reached already in 2021. The planned improvement in the budget balance is backloaded, as a deterioration of about 0.3 pps. of GDP is projected in 2019. All Member States that notified deficits in 2018 plan to improve the headline position by 2022, except EE and PL. Member states that had reached a surplus in 2018 often plan to reduce it (except SI, HR, EL and SE), some of them to a balanced budget (NL, DK, AT) or even a deficit (CZ).



Note: The graph shows the headline budget balance in 2018 and 2022 according to the 2019 SCPs, and the change planned between 2018 and 2022. For CY the change in the budget balance (+6.9 pps. of GDP) falls outside the scope of the graph. **Source:** European Commission based on 2019 SCPs

The slight increase in the aggregate headline balance over the programme horizon results from a planned reduction in the primary expenditure ratio, which more than offset the decline in the revenue ratio (Graph 2.2). Specifically, the primary expenditure ratio is planned to decline by around 0.7 pps. of GDP in both the EU and the euro area by 2022. This decline is driven by positive cyclical conditions, as real GDP growth is projected to outpace (medium-term) potential growth. The projected drop in interest expenditure (of around 0.2 pps. of GDP between 2018 and 2020) would also contribute to the fiscal adjustment. At the same time, the revenue ratio is projected to fall by around 0.5 pps. of GDP, with only 0.2 pps. explained by the reported discretionary measures. This could imply that further revenue decreasing measures need to be specified.





Note: A positive sign indicates a contribution to the fiscal adjustment. In other words, a drop in primary expenditure or interest expenditure is shown as a positive value, while a drop in revenues is shown as a negative value. **Source:** European Commission based on 2019 SCPs

Looking at the planned evolution of the (recalculated) structural balance (¹⁴), Member States' plans would result in a worsening of the aggregate structural balance in 2019, followed by a slight fiscal tightening in 2020-2022 (Graph 2.3). These developments show significant revisions of the plans compared to the 2018 SCPs. Specifically, the structural balance is expected to worsen by around 0.2 pps. of GDP in both the EU and the euro area in 2019, while it was planned to improve by 0.3 pps. of GDP in last year's programmes. The planned structural improvement has been more than halved in 2020-2021 compared to the 2018 SCPs, from 0.7 pps. of GDP to 0.3 pps. in the EU and 0.2 pps. in the euro area. In 2022, the structural balance is planned to improve, however, remain in deficit, at 0.5% of GDP in the EU and 0.6% in the euro area. A detailed discussion of the fiscal stance in the euro area, based both on the Commission 2019 spring forecast and on the 2019 SCPs, is provided in subsection 2.2.

For euro area Member States, a comparison of 2019 macroeconomic and budgetary projections between the SPs and the DPBs can be found in Box 2.1.

Most Member States plan to move in the direction of their MTO or to remain at or above it (Graph 2.4). All Member States that were below their MTO in 2018 plan to increase their structural balance by 2022, except EE. (¹⁵) In the 2019 SCPs, Member States have set their MTO for 2020-2022 in line or above the new minimum MTO (Box 2.3). The planned structural effort in some Member States that had high structural deficits in 2018 (ES, FR, IT and RO) is not sufficient to reach their MTO by 2022, while HU plans to achieve the MTO. All Member States at or above the MTO in 2018 intend to remain above it in 2022, though a majority of them plan a fiscal expansion over the SCP horizon. By the end of the programme horizon, 23 Member States plan to have reached the MTO or to be in its vicinity, (within 0.25% of GDP distance) – see also Box 2.2.

^{(&}lt;sup>14</sup>) Recalculated by the Commission based on the information provided in the programmes according to the commonly agreed methodology.

^{(&}lt;sup>15</sup>) This is based on the recalculated structural balances. For EE, at face value the Stability Programme shows an improvement in the structural deficit from 1.4% of GDP in 2018 to 1.0% in 2019, 0.8% in 2020-2022 and 0.1% in 2023. Anyway, the Stability Programme is based on unchanged policy.

Box 2.1: Comparison with the Draft Budgetary Plans of euro area Member State

The cycle of fiscal surveillance for euro area Member States is structured around two main milestones in the calendar year: the submission of Stability Programmes (SPs) in spring and the submission of Draft Budgetary Plans (DBPs) in autumn. While the latter concerns only the following year, the SPs contain the authorities' medium-term fiscal strategies. However, the two fiscal documents overlap for one year, 2019 in this case. The information contained in the SPs for 2019 is thus an update of Member States' plans as presented to their euro area peers last autumn. This box compares the macroeconomic outlook and budgetary targets for 2019, as presented in the most recent DBPs and SPs. In the case of LV, LU and SI, data from a no-policy-change DBPs as submitted in October 2018 is used.

The 2019 growth outlook for the euro area has been revised downwards since last autumn (Table 1), with real GDP growth now projected at 1.4% compared to 2.0% in the DBPs. The GDP deflator is now expected at 1.4% in 2019 (compared to 1.8% in the DBPs). The downward revision of the growth outlook is accompanied by a slight downward revision in the change of the headline balance (-0.2 pps. of GDP). The planned fiscal effort based on the expenditure benchmark is now slightly more negative (to -0.4 pps. from -0.3 pps. of GDP), while the change in the structural balance is similar.

			Change	Change	Fiscal effort		
	Real GDP	GDP deflator	in	in			
	growth	growth	Headline	structural	based on		
	8	8	balance	balance	EB		
BE	-0.2	0.0	-0.2	-0.1	0.3		
CY	-0.2	-0.3	7.6	-0.2	-3.6		
DE	-0.8	0.4	-0.2	-0.1	0.0		
EE	0.1	-0.4	0.4	0.4	0.8		
EL	-0.2	-0.2	0.3	0.1	-1.1		
IE	-0.3	-0.4	0.1	0.8	0.2		
ES	0.0	-0.2	-0.5	-0.5	-0.8		
FR	-0.3	-0.1	-0.4	0.0	-0.4		
IT	-1.3	-0.6	0.3	0.7	0.3		
LV	0.2	0.0	0.4	0.4	-0.6		
LT	-0.2	-0.3	-0.1	0.0	0.2		
LU	-1.0	-0.2	-1.3	-1.3	0.0		
МТ	0.9	0.7	-1.3	-1.2	-0.6		
NL	-1.1	-0.3	-0.5	0.1	0.1		
AT	-0.3	-0.1	-0.2	-0.2	-0.3		
РТ	-0.3	0.1	-0.2	0.0	0.0		
SI	-0.3	0.4	0.6	0.9	0.4		
SK	-0.5	0.1	0.2	0.3	-0.4		
FI	0.0	-0.1	-0.3	-0.2	-0.6		
EA	-0.6	-0.3	-0.2	0.0	-0.1		

Table 1: Revision of selected indicators in 2019: SPs vs DBPs

Note: The table shows the amount of revisions in real GDP and deflator growths (pps.) and in selected fiscal indicators (pps. of GDP). In CY, the planned improvement in the headline balance has been raised from 0.2 pps. of GDP in the DBP to 7.8 pps. in the SP due to large one-offs in 2018.

Source: European Commission based on 2019 SPs and 2019 DBPs



Note: The graph shows the change in the structural balance: in 2019, 2020, over the period 2021-2022 and over the entire period 2019-2022, according to the 2019 SCPs. Countries are ordered from the smallest to the largest cumulative change in the structural balance over the entire period 2019-2022. Countries in the circle were at or above the MTO in 2018. The structural balance in EL was 5% of potential GDP in 2018; however the country has nominated its first MTO after exiting the ESM Stability Support Programme in the 2019 SC +(+0.25%; valid from 2020). **Source:** European Commission based on 2019 SCPs



Notes: The graph sets out Member States' reported progress towards their respective MTOs throughout the programme horizon. The base of the arrow represents the starting position of the structural balance in 2018 based on the SCPs, while the tip of the arrow represents the projected (recalculated) structural balance in 2022. Thus, the size of the green (red) arrow corresponds to the projected overall improvement (deterioration) in the structural balance throughout the programme horizon. Finally, the dark squares represent each Member State's MTO set for 2020 that is assumed to remain unchanged until 2022. For the euro area and the EU, an aggregate 'MTO' were calculated as weighted averages of country-specific MTOs. Member States are ranked according to their distance to the MTO in 2018.

Source: European Commission based on 2019 SCPs



The number of Member States close to or above the MTO increased strongly over the period 2011-2018, yet 10 Member States have not reached the MTO a single time. The number of Member States close to or above the MTO increased from 3 in 2011 to 12 in 2018, with the peak of 14 in 2016 (Graph 1a). While 12 Member States were at least half of the time close to or above the MTO over the period 2011-2018, 10 Member States had not reached the MTO a single time (Graph 1b).

Graph 1: Progress towards the MTO



Notes: EL is included in the graph from 2020 onwards. Member States are considered to be close to the MTO if the distance to it does not exceed 0.25% of GDP. **Source:** European Commission and 2019 SCPs.

According to 2019 SCPs, the number of Member States close to or above the MTO is planned to increase from 14 in 2019 (Graph 2a) to 23 by 2022 (Graph 2b). The progress towards the MTO is planned to be uneven across Member States. Member States that reached or exceeded the MTO at least once over 2011-2018 plan to be close to or above the MTO at least three times over the period 2019-2022 (except HU, RO and EE). Conversely, three large Member States that were not a single time at the MTO over 2011-2018, plan to remain below the MTO also over the programme horizon (ES, FR, IT).



(Continued on the next page)



Notes: In panel b), the two groups of Member States are formed according to the structural balance position vis-a-vis the MTOs in 2018 using the unweighted average. Member States that are included into the green group (i.e. at or above the MTO in 2018) are BG, CZ, DE, EL, HR, CY, LT, LU, MT, NL, SE, DK and AT, while Member States in the red group (i.e. below the MTO in 2018) are PL, FI, IE, EE, SK, BE, HU, SI, UK, FR, IT, PT, RO, ES and LV. **Source:** European Commission and 2019 SCPs

Member States that were at the MTO in 2018 have gradually increased the gap in the structural balance position compared to the groups of Member States that were not at the MTO in 2018 (Graph 3b). This group achieved most of the structural effort over 2011-2013 and 2015-2017. Member States that were below the MTO in 2018 more than halved the distance to the MTO between 2011 and 2013. Since then, their progress towards the MTO has been limited or absent. According to the SCPs, both groups plan to be at least close to the MTO by the end of the SCP horizon. The strongest improvement in the structural balance is planned in the group below the MTO in 2018 while the group above the MTO plans to reduce the structural balance, yet remain well above the MTO.

Box 2.3: Update of the minimum MTO

The minimum MTOs were recently updated based on the new set of long-term budgetary projections published in 2018. (¹) According to the Regulation (EC) 1466/97, the minimum MTO should be revised every three years, after the publication of the "Ageing Report" that provides up-to-date data on the ageing challenge facing Member States. The recently updated minimum MTOs will apply for the period 2020-2022. Compared to the current set of minimum MTOs, the new minimum MTO calculation yields the same outcome for 10 Member States (AT, CY, DK, ES, FI, FR, LT, LV, NL and SK) - Graph 1. For 10 Member States the new values are more stringent. This is either due to higher costs of ageing (BE, BG, CZ, IT, LU, HU and UK) or the increase in the minimum benchmark (EE, HR and RO). For 7 Member States the minimum MTOs are less stringent (DE, IE, MT, PL, PT and SI and SE) due to the lower debt ratios. In the case of EL, the computation of the minimum MTO for 2020-2022 is the first after the country exited the ESM Stability Support Programme.

Nine Member States changed the MTO for 2020-2022, eight as a response to the minimum MTO update. The minimum MTO represents a lower bound for the MTO that Member States have to respect in order to be considered compliant with the SGP. Member States nominated their MTOs for the period 2020-2022 in the 2019 Stability and Convergence Programmes (SCPs). Five Member States (UK (²), HU, CZ, IT and LU) nominated stricter MTOs due to the upward revision of the minimum MTOs and one Member State (HR) increased its MTO, also to fulfil the ERMII entry condition. The downward revision of the minimum MTO had an impact on two Member States (SI, PT) while one Member State reduced the MTO to the minimum MTO despite no change in the updated minimum MTO (SK).

Graph 1: Update of the minimum MTO



Notes: Member States are ranked according to the change in the minimum MTO, from the largest reduction to the largest increase. The computation of EL minimum MTO for 2020-2022 is the first after EL exited the ESM Stability Support Programme. EL nominated for the first time its MTO in 2019 Stability Programme. **Source:** European Commission based on 2019 SCPs

(¹) The 2018 Ageing Report: Economic and Budgetary Projections for the EU Member States (2016-2070).
 (²) The UK has never set in the Convergence Programmes its MTO so the minimum MTO has always been considered.

At aggregate level, the planned developments in the expenditure benchmark point to a negative fiscal effort in 2019, followed by a neutral stance in 2020-2022 (Graph 2.5). The projections for primary expenditure - net of discretionary revenues - put forward in the programmes imply a negative fiscal effort in both the EU and in the euro area in 2019 (of -0.3 pps. and -0.4 pps. of GDP respectively). The main negative contribution comes from the planned increase in primary expenditure outpacing medium-term nominal potential growth. At the same time, the SCPs project some discretionary tax cuts in 2019 (Graph 1.6). In 2020, the fiscal effort based on the expenditure benchmark is projected at zero in

both the EU and the euro area, as the planned increases in taxes would offset the negative impact of primary expenditure growing more than medium-term potential growth. In 2021-2022, the planned developments in primary expenditure and discretionary revenue reverse again, but the expenditure benchmark continues to point to a fiscal effort close to zero in both the EU and the euro area thanks to the primary expenditure restraints offsetting the planned tax cuts.

At country level, nine Member States plan a positive fiscal effort in 2019, based on the expenditure benchmark (Graph 2.5). However, the expenditure benchmark indicates a planned negative fiscal effort, or no effort, in some highly-indebted countries that have not achieved the MTO (ES, FR, IT and PT) as well as in a high-deficit country like RO. For 2020, the expenditure benchmark points to a positive fiscal effort in 14 Member States, but the fiscal effort is planned to be slightly negative in ES, FR and PT. Over the entire 2019-2022 programme period, 12 Member States plan a positive fiscal effort based on the expenditure benchmark, with the largest positive fiscal effort projected in IE and SK.



Note: in 2019, for CY, the difference between the fiscal effort based on the expenditure benchmark (-3.7% of GDP) and the change in the structural balance (-0.8 pps. of GDP) is largely explained by sizeable one-offs that increased Gross Fixed Capital Formation (GFCF) in 2018. While in both indicators one-offs are netted out in the year in which they occur (2018 in this case), in the expenditure benchmark GFCF is smoothed over four years implying lower expenditure net of one-offs in 2018 and higher in 2019 compared with those used in the structural balance. **Source:** European Commission based on 2019 SCPs

Based on the 2019 SCPs, the aggregate debt-to-GDP ratio would decline by around 7 pps. between 2018 and 2022 (Graph 2.6). The EU public debt would fall to around 75% of GDP in 2022, while for the euro area the debt ratio would be around 80%. The decline in the debt ratio would be driven by the planned primary surpluses and a favourable growth-interest differential, resulting in a debt-reducing snowball effect. At country level, debt reductions larger than 10 pps. of GDP are planned in 10 Member States (EL, CY, PT, SI, MT, HR, AT, IE, HU and SE). A large decline in the debt ratio is also projected in DE and NL. A debt-decreasing snowball effect is projected in all Member States but Italy, while the accumulation of primary deficits would have a debt-increasing impact in 3 Member States (FR, RO and EE). The number of Member States with a debt ratio above 60% of GDP would fall from 14 in 2018 to 10 in 2022. Among the 10 countries that project a debt above 60% of GDP in 2022, FR and IT do not plan to reduce the gap to the 60% ceiling by at least 1/20th per year on average over the programme period. In 2022, debt ratio would remain above 100% of GDP in EL (153%), IT (129%) and PT (104%).

All Member States with a debt ratio above 60% of GDP and not at the MTO in 2018 plan some structural adjustment (Graph 2.7), although such adjustment is not always correlated to the size of the debt ratio.





Note: The graph disaggregates the 2019-2022 change to Member States debt-to-GDP ratios between the contributions of the primary balance, stock-flow adjustments and the snowball-effect (i.e. due to the interest rate-growth rate differential). Stock-flow adjustments are calculated as the residual between the annual changes in the debt levels of the SCPs and the reported headline balances. Values below (above) zero indicate a decreasing (increasing) impact on the debt ratio. **Source:** European Commission based on 2019 SCPs





31

2.2. ASSESSMENT OF THE EURO AREA FISCAL STANCE

This section analyses the euro area fiscal stance in the context of the broader policy mix. Specifically, Section 2.2.1 describes the euro area fiscal stance in 2019 based on the Commission 2019 spring forecast and the monetary policy stance. Section 2.2.2 discusses of the appropriate fiscal stance for the euro area in 2020, based on cyclical stabilisation and medium-term sustainability needs. Section 2.2.3 assesses the fiscal stance emerging from the Stability Programmes.

2.2.1. Description of the monetary policy stance and fiscal stance in 2019

Monetary policy in the euro area remains highly accommodative (Graph 2.8). At its March and June meetings, the ECB Governing Council reacted to the continued economic slowdown and further declines in medium-term inflation expectations. Although policy rates were left unchanged, the ECB extended the date-contingent part of its rate forward guidance by one year over the two meetings, now expecting that key ECB interest rates "remain at their present levels at least through the first half of 2020 [...]". As a result, EONIA forward-based expectations of a first ECB policy rate hike were pushed out considerably, and in fact markets now even price in a rate cut over the coming months. At the time of writing they implied 10 bps lower rates by September 2019 and a return to current levels only by 2022. While the forward guidance on full reinvestments of maturing APP securities was kept unchanged, its link to a first policy rate hike(16) implies a parallel extension of reinvestment expectations, ensuring that the APP effects on long-term yields stay in place. The Governing Council furthermore in March announced a new series of quarterly targeted longer-term refinancing operations (TLTRO-III), starting in September 2019 and ending in March 2021, each with a maturity of two years. With respect to the policy outlook, the Governing Council renewed its pledge that it "stands ready to adjust all of its instruments, as appropriate, to ensure that inflation continues to move towards the Governing Council's inflation aim in a sustained manner" and emphasized its determination to act in case of adverse contingencies.



^{(&}lt;sup>16</sup>) The forward guidance on reinvestments states that they are expected to continue in full "for an extended period of time past the date when [the Governing Council] starts raising the key ECB interest rates"

The accommodative monetary policy stance is expected to keep financing conditions favourable in the euro area. Developments in credit costs have been overall very muted in the euro area over the past quarters, with the composite credit cost indicators (CCCI) for households and non-financial corporations remaining near their all-time lows. While credit costs for households have moved sideways for the past year, non-financial corporations' credit costs actually increased slightly until January. This reflected offsetting movements by medium- and long-term loans, which decreased in line with long-term money market rates, and increasing corporate bond yields due to higher risk premia. This upward trend, however, was reversed thereafter, when corporate bond yields declined considerably, resulting in a broadly unchanged CCCI for non-financial corporations compared to mid-2018. Looking ahead, ECB forward guidance on policy rates suggests low short-term rates for some time, while the associated continued reinvestment of maturing securities purchased under the Expanded Asset Purchase Programme (EAPP) in combination with the high overall stock of assets on the Eurosystem's balance sheet should ensure that nominal long-term rates to pick up only modestly over the coming two years and to remain well below levels reached at the end of 2018.

The Commission 2019 spring forecast projects an expansionary euro area fiscal stance in 2019. More specifically, in 2019 the euro area fiscal stance appears to be expansionary when looking at the DFE, with an overall fiscal impulse of around ½ pp. of GDP (Graphs 2.9). The expansionary fiscal impulse from primary expenditure – net of one-offs and cyclical unemployment benefits – would be around ¼ pp. of GDP in 2019, as it is set to increase more than medium-term nominal potential growth (by 3.4% as against 2.8% - Table 1.3). A similar fiscal impulse is expected on the revenue side due to discretionary tax cuts (Graph 1.6). The worsening of the structural balance, by around ¼ pp. of GDP, points to a mildly expansionary fiscal stance in 2019 (Graph 1.5). (¹⁷) This result is confirmed also by the ECB (Economic Bulletin, Issue 4, 2019) The combination of accomodative monetary policy and slightly expansionary fiscal stance is supporting the euro area economy in a context of a weaker external demand.

In terms of geographical composition, most euro area countries would run an expansionary fiscal stance based on the DFE in 2019. The DFE points to an expansionary fiscal stance together with a positive output gap in 10 euro area Member States (Graph 2.9). Looking at the 5 largest countries, the fiscal stance is projected to be broadly neutral in France, slightly expansionary in Italy and expansionary in Spain, Germany and the Netherlands. (¹⁸) The fiscal expansion in Germany appears to be counter-cyclical, given that, despite the fiscal expansion, real GDP growth is forecast to be significantly lower than potential growth and the output gap is estimated to turn marginally negative.

^{(&}lt;sup>17</sup>) Around half of the ¹/₄ pp. of GDP difference between the DFE and the change in the structural balance in 2019 is explained by the projected revenue windfalls in the Commission 2019 spring forecast. The remainder is mainly due to the lower real potential growth used in the DFE (1.2% as against 1.3%).

^{(&}lt;sup>18</sup>) Based on the structural balance, in 2019 nine euro area countries would run a broadly neutral fiscal policy stance, as their change in the structural balance would fall in a range between -0.2 pps. and +0.2 pps. of potential GDP, combined with a positive output gap. As for the five largest euro area Member States, an expansionary fiscal policy would be confirmed in Germany, with a fiscal impulse of almost ½ pp. of GDP, and a broadly neutral stance in FR also based on the change in the structural balance. Fiscal policy would be broadly neutral in Spain, the Netherlands and Italy. It should be noted that in the Commission 2019 spring forecast Germany's potential growth for 2019 has been revised to 1.5%, from 2.1% estimated in autumn 2018. This revision implies, other things being equal, a worsening in the change of the structural balance of around ¼ pp. of GDP in 2019.



Note: In a context of positive output gaps, 'pro-cyclical' and 'counter-cyclical' refer in this graph to whether the fiscal policy stance contributes to support or to drag on the economy, CY and EL fall outside the graph with a DFE of -4% and -0.7% of GDP and output gap of +3.8% and -4.0% of potential GDP, respectively. **Source:** European Commission 2019 spring forecast

2.2.2. What could be an appropriate euro area fiscal stance in 2020?

Fiscal policy should ensure public finance sustainability while supporting macroeconomic stabilisation in the short term, in particular when monetary policy is constrained at the zero lower bound. (¹⁹) These objectives are presented through a 'fiscal map' that uses the S1 indicator to measure medium-term sustainability needs (²⁰) and the output gap to measure cyclical stabilisation needs (Graph 2.10).

^{(&}lt;sup>19</sup>) This follows the approach presented in European Commission (2016). "The fiscal stance in the euro area: Methodological issues". *Report on Public Finances in EMU*.

^{(&}lt;sup>20</sup>) The S1 indicator measures the change in the structural primary balance (dSPB) required over five years (from 2020 to 2024) to bring general government debt to the reference threshold of 60% of GDP in 2033. The S1 indicator represents a stylised measure of sustainability needs. In the Commission fiscal sustainability assessment framework, the assessment of medium-term sustainability risks is based on a broader approach including a Debt Sustainability Analysis, encompassing a large set of deterministic and stochastic projections (Box 4.1 for more details and references). In addition, the measure of sustainability needs provided by the S1 indicator can differ from the perspective of the compliance to the SGP (and therefore the Country-Specific Recommendations). A negative S1 indicates the presence of fiscal scope.


Graph 2.10: Fiscal map: Sustainability and stabilisation challenges in 2020 (% of GDP)

Notes: Horizontal axis: indicator of medium-term risk to the sustainability of public finances: \$1 scenario 2019, which measures the total cumulative adjustment (in terms of change in the structural primary balance over 5 years) that would be needed to bring the debt ratio to 60% of GDP by 2033. Vertical axis: Output gap for 2019 based on the commonly agreed methodology. The size of the bubbles is related the level of GDP. **Source:** European Commission 2019 spring forecast

On the one hand, concerns about fiscal sustainability call for rebuilding fiscal buffers in highlyindebted countries in 2020. Five euro area Member States (Belgium, Spain, France, Italy and Portugal) show substantial sustainability needs in 2020, based on the S1 indicator for 2019 (Graph 2.10 and Graph 2.11). At the same time, according to the Commission 2019 spring forecast, no euro area Member State is expected to experience bad economic conditions based on the output gap. (²¹) In 2019, the output gap would be only marginally negative in Germany and Italy and positive in the rest of the euro area, while in 2020 economic growth is set to return marginally above potential growth in a context of a very accommodative monetary policy. As a result, no stabilisation needs seem to emerge for 2020. Hence, fiscal sustainability concerns should prevail in those Member States where sustainability risks are high and public debts are not on a safe downward trajectory. Reducing large structural deficits should be the priority for highly-indebted countries, as in the event a sharp economic downturn a stronger fiscal position would allow them to let the automatic stabilisers play in full their role. By contrast, a weak fiscal position may trigger financial market reactions and force pro-cyclical contractionary fiscal policies entailing high social costs.

^{(&}lt;sup>21</sup>) Greece is not included in this analysis, as S1 data are not available.



Graph 2.11: Change in the structural primary balance in 2020: Stability Programmes' plans against sustainability and stabilisation needs (% of GDP)

Notes: The sustainability criterion is based on the \$1 indicator for 2019 and assumes that 20% of the needed change in the structural primary balance (SPB) is implemented in 2020 (red dot). For countries with a negative \$1, this indicates scope for expansionary policies. The stabilisation criterion is measured as the change in the SPB for which in 2020 fiscal policy would reduce the output gap expected in 2019 by 20% (blue diamond). This assumes that fiscal policy always plays a countercyclical role (with a fiscal multiplier of 0.75) supporting the closure of the output gap, while other changes in the SPB in 2020 presented by Member States in their 2019 Stability Programmes, as recalculated by the Commission using the commonly agreed methodology for potential output.

Source: European Commission based on Commission 2019 spring forecast and 2019 Stability Programmes

On the other hand, member States with fiscal scope could boost their potential growth taking advantage of the low interest rate environment. Countries with a negative S1 indicator could take advantage of interest rates remaining at their historical lows to increase their future growth prospects by investing in public infrastructure and other productive spending. It could be in fact argued that, in an economic environment where the cost of servicing the debt may remain significantly below potential economic growth for a protracted period (i.e. r<g), reducing public debt significantly below 60% of GDP might not be a priority in countries experiencing physical and digital infrastructure gaps. (22) In terms of fiscal support to the economy, the full use of automatic stabilisers in countries with fiscal space might be sufficient to counter the central scenario of a mild slowdown. However, additional productive spending in those countries would also help to reduce the negative impact on the economy in case the substantial downside risks surrounding economic growth projections for 2020 materialised (Graph 2.14). Given that monetary policy is at its zero lower bound, fiscal multipliers and spillovers to the rest of the euro area would be rather large. (²³) As a result, a well-designed fiscal expansion in euro area Member States with fiscal scope would enhance their short- and medium-term growth prospects while supporting the overall euro area economy at a time when highly-indebted countries have to reduce their sustainability needs. Moreover, higher public investment would help to rebalance the economy of some Member States like

^{(&}lt;sup>22</sup>) See for instance Blanchard, O. (2019). "Public Debt and Low Interest Rates", AEA lecture January 2019. Implicit interest rates paid on public debt are currently lower than GDP growth in all euro area Member States except Italy, so r<g implies debt-decreasing snowball effects. However, as seen in the case of Italy, negative r-g values cannot be taken for granted in all Member States. Policies that are perceived as unsustainable may increase the cost of servicing the debt with limited or even negative impact on growth prospects.</p>

^{(&}lt;sup>23</sup>) See for instance In 't Veld, J. (2016). "Public investment stimulus in surplus countries and their euro area spillovers", *Economic Brief 16*, August 2016, <u>https://ec.europa.eu/info/sites/info/files/file_import/eb016_en_2.pdf</u> and Buti, M. Leandro, J. and K. Berti (2017). "An unusual recovery: Charting the way forward for European policymakers", *EU Vox*, 12 May.

Germany and the Netherlands toward domestic demand and possibly help reduce large domestic savingsinvestment imbalances in the euro area (Box 2.7).

Therefore, differentiated fiscal policies would be advisable in the euro area in 2020: (24)

- Member States with high public debt should strengthen their fiscal position. Countries that have limited or no fiscal scope, as also shown by the sizeable sustainability needs based on the S1 indicator (Graph 2.11) should implement fiscal adjustments. The accommodative monetary policy stance would support them in their efforts. However, based on unchanged policies, in 2020 the Commission forecast indicates that the structural deficit would remain broadly stable in France and worsen in Belgium, Spain and especially Italy (by more than 1 pp. of GDP). Italy is also the only euro area country where the debt ratio is projected to increase further over the forecast horizon, while debt reduction would be small in Belgium, Spain and France. It should be recalled that in 2018 the announcement of large expansionary policies triggered a significant increase in interest rates in Italy. The overall impact on growth of these expansionary policies could have actually been contractionary because of worsening financing conditions also for the private sector and increased uncertainty about the sustainability of public finances. (²⁵)
- Member States with fiscal scope may use (part of) it to increase productive spending. Euro area countries that have fiscal scope, as also indicated by the negative sustainability needs based on the S1 indicator, may use (part of) it to support growth-enhancing spending. A fiscal expansion that focused on productive spending would boost potential growth and thus reduce possible overheating concerns where stabilisation needs would suggest restrictive fiscal policies as appropriate (Graph 2.11). Additional productive spending in countries with fiscal scope would be also an insurance for them and the rest of the euro area against the substantial downside risks to the economic outlook that, if materialised, may also imply deflationary effects. In 2020, there is space for using fiscal policies to boost investment and other productive spending while respecting the MTO in six euro area Member States. Based on the Commission no-policy-change forecast, Germany is projected to implement a mild fiscal expansion in 2020 by reducing its structural surplus by around ¹/₄ pp. of GDP, while in the Netherlands the fiscal expansion would be of around ¹/₂ pp. of GDP.

The appropriate differentiation of fiscal policies across member States would lead to a broadly neutral fiscal stance at the euro area level in 2020. The combination of restrictive fiscal policies in Member States that are required to adjust (26), in line with the Country-Specific Recommendations (CSRs) for 2020, and additional productive spending in Member States that are expected to overachieve their MTO in 2019, leading to decline in their structural balance up to $\frac{1}{2}$ pp. of GDP), would lead to a broadly neutral euro area fiscal stance in 2020. The outcome of these differentiated fiscal policies is shown in the "SGP compliance and use of fiscal space" scenario in Graph 2.12 for the change in the structural balance and Graph 2.13 for the increase in net primary expenditure relative to medium-term nominal potential growth (27)

Based on unchanged policies, the Commission 2019 spring forecast points to a slightly expansionary fiscal stance for the euro area aggregate in 2020, which would be pro-cyclical. Based on the DFE, the

 $^(^{24})$ It should be stressed here that the economic considerations drawn from the fiscal map that follow can in no way dispense Member States from their obligations under the SGP.

⁽²⁵⁾ See for instance Blanchard O. and Zettelmeyer J. (2018). "The Italian Budget: A Case of Contractionary Fiscal Expansion?". *Real Time Economic Issues Watch PIIE*, <u>https://piie.com/blogs/realtime-economic-issues-watch/italian-budget-case-contractionary-fiscal-expansion</u>

⁽²⁶⁾ The eleven euro are Member States with a required fiscal effort in 2020 are BE, EE, IE, ES, FR, IT, LV, PT, SI, SK and FI.

^{(&}lt;sup>27</sup>) The "SGP compliance and use of fiscal space" scenario implies an increase in the euro area structural balance close to 0.15 pps. of GDP in 2020 (Graph 2.10). Euro area net primary expenditure would instead increase by 2.6%, i.e. slightly less than the 2.9% increase in medium-term nominal potential growth estimated by the Commission forecast (Graph 2.11). This scenario compares with a "SGP compliance" scenario, which assumes full compliance with CSRs in countries required to implement a fiscal adjustment and no change in the structural balance of around 0.35 pps. of GDP, while net primary expenditure would rise by 2.1%).

fiscal expansion would be 0.4 pps. of GDP, while the structural balance is projected to worsen by 0.3 pps. However, as mentioned above, the geographical composition of this aggregate fiscal stance would be suboptimal, in particular because, based on unchanged policies, some highly-indebted Member States would worsen their structural position. It may be argued that at the current juncture the actual impact on economic growth of expansionary policies in the euro area depends more on the geographical composition than on the aggregate size of the fiscal impulse. The demand effects of fiscal expansion in highly-indebted countries should in fact be weighed against the risk of tighter financing conditions, with uncertainty about the sign of the net fiscal multiplier of expansionary policies in some of those countries. According to the 2019 Stability Programmes, the euro area fiscal stance would be broadly neutral in 2020 as some Member States with sustainability needs pledge to adopt consolidation measures.



Note: The scenarios displayed in the graph represent the change in the aggregate structural balance of the euro area. The scenario 'SGP compliance' assumes that Member States that are not at their MTO implement the fiscal adjustment recommended by the Country-Specific Recommendations for 2020, while no change in the structural balance is assumed in Member States that are projected to overachieve their MTO in 2019. The scenario 'SGP compliance and use of fiscal space' assumes on top of SGP compliance that Member States that have overachieved their MTO in 2019 use (part of) their fiscal scope in 2020 with a decrease in the structural balance up to 0.5 pps. of GDP, depending on the distance to the MTO. The 'Member States plans' and the 'SGP compliance and use of fiscal space' scenarios give similar results for the euro area aggregate, but the composition at country level is different and the former is not consistent with the SGP for some Member States.

Source: European Commission based on Commission 2019 spring forecast and 2019 Stability Programmes



Graph 2.13: Fiscal stance scenarios in 2020 based on the % change in net primary expenditure

Note: The scenarios displayed in the graph represent the increase in net primary expenditure (based on the expenditure benchmark) of the euro area compared with 10-year potential growth estimates. The scenario 'SGP compliance' assumes that Member States that are not at their MTO implement the increase in net primary expenditure as recommended by the Country-Specific Recommendations for 2020, while no change in the structural balance is assumed in Member States that are projected to overachieve their MTO in 2019. The scenario 'SGP compliance and use of fiscal space' assumes on top of SGP compliance that Member States that have overachieved their MTO in 2019 use (part of) their fiscal scope in 2020 with a decrease in the structural balance to the MTO. The 'Member States plans' and the 'SGP compliance and use of fiscal space' scenarios give similar results for the euro area aggregate, but the composition at country level is different and the former is not consistent with the SGP for some Member States. **Source:** European Commission based on Commission 2019 spring forecast and 2019 Stability Programmes

A call for an expansionary fiscal stance in the euro area does not seem to be warranted at this stage. This would require a dramatic change in the economic outlook. However, Graph 2.14 suggests that the balance of risks for euro area economic growth projections is tilted to the downside. Of course, if those downside risks were to materialise, the situation would need to be reassessed. Anyhow, Member States with sustainability challenges would have limited room to undertake expansionary fiscal policy. In those countries, the overall impact on growth of expansionary policies could be limited by increased uncertainty about the sustainability of public finances and possibly worsening financing conditions. By contrast, in case of a sharp economic downturn, countries with fiscal space should do more to cushion it.

Based on alternative indicators, sustainability needs may be lower than shown by the S1 indicator, but stabilisation needs would remain limited. Specifically, the current level of unemployment does not seem to indicate any larger stabilisation need compared to the output gap (Box 2.4). At the same time, the sustainability needs would appear less severe if computed with a debt threshold of 90% of GDP (instead of 60% as in the S1) and interest rate assumptions based on current financial markets' expectations (Box 2.5).

Finally, structural policies should be an integral part of the right policy mix to support growth prospects and the resilience of the euro area. In particular, fiscal structural policies that prioritise private investment and other productive spending while reducing the tax burden on labour and capital would have positive impact on growth prospects (Box 2.6).





2.2.3. Assessing the fiscal stance derived from the Stability Programmes

The 2019 Stability Programmes points to a broadly neutral fiscal stance for the euro area in 2020. The euro area structural balance is planned to increase by slightly more than 0.1 pp. of GDP (Graph 2.12) in 2020, in line with the structural effort implied by the planned increase in net primary expenditure (Graph 2.13). Booth indicators thus point to a broadly neutral euro area fiscal stance planned by Member States in 2020.

Member States' plans are not always consistent with sustainability needs, as also reflected more institutionally in the lack of compliance with the CSRs:

- For most countries, the changes in the structural primary balance planned for 2020 in the Stability Programmes are larger than their sustainability needs, as measured by a 20% reduction of the medium-term sustainability indicator S1 (Graph 2.11) as well as the CSRs. The planned fiscal expansion in Germany is consistent with its sustainability needs, while the Netherlands do not plan any fiscal expansion despite having fiscal scope.
- Some Member States (Belgium, France, Portugal and Italy) plan fiscal efforts that fall short of their large sustainability needs (and the CSRs). Moreover, for Belgium, Spain and especially Italy the change in the structural primary balance projected at unchanged policy in the Commission 2019 spring forecast is significantly lower than the Stability Programme target for 2020, implying the need of sizeable consolidation measures.

Box 2.4: Stabilisation needs in the euro area as identified by a double condition on unemployment

In its proposal for a European Investment Stabilisation Function, the Commission suggested that a 'double condition' on the unemployment rate would indicate the presence of a large cyclical shock. The unemployment rate provides a robust indicator of business cycle movements that leaves aside high-frequency fluctuations. It is also simple, subject to small revisions and its measurement is harmonised. The proposed double condition is specified as (i) the quarterly national unemployment rate exceeding its average over a period of 60 quarters (15 years), and (ii) a year-on-year increase in the quarterly unemployment rate by more than 1 percentage point. An unemployment rate above its historic average neutralises differences in permanent (structural) level of unemployment. (¹) A marked annual increase in unemployment rate together with a level higher than average sends an unambiguous signal of weak cyclical conditions Thus, the proposed criterion is highly unlikely to flash out in the absence of stabilisation needs (i.e. risk of false positives). However, the double criterion may not be met in *any* possible situation where a stabilisation need could be argued. (²)

What is the signal from the double condition at the current juncture?

Labour market conditions have steadily improved in recent years. Since the start of the recovery over five years ago, unemployment rates have been on a downward trend. Despite the slowing economic activity, labour market conditions have continued to improve in recent quarters and the euro area unemployment rate fell in 2019Q1 to its lowest level since autumn 2008. In 2019 and 2020, unemployment rates are forecast to decline more slowly in the euro area, with the exception of Malta and Italy where an increase is projected.

In line with these evolutions, the double condition is not met now in any Member State (Table 1). In three Member States only (LU, EL, IT), the unemployment rates are above their long-term average (hence, the 'level condition' is met). However, unemployment rates are still decreasing in all countries on a yearly basis (the 'change condition' is not met). In general, most Member States are relatively far from meeting the double condition at this point, given the very significant labour market improvements from past years, as illustrated by large falls from past peaks.

To what extent would the double condition be met in the event of a shock?

As noted above, the baseline forecast is one of a continued decline in unemployment rates (or only very small increases in few Member States). Should downside risks to the forecast materialise however, it is conceivable that the labour market turns around. The double condition may then be met at some point at least in some Member States.

To illustrate to what extent this could occur, three scenarios based on historical experience are simulated, reflecting different degrees of intensity of the negative shocks:

- i) A relatively moderate slowdown, as experienced in the early 2000s, whereby the euro area unemployment rate rose by close to 1pp cumulated over 2002-2004;
- ii) A more significant downturn, of the kind experienced in the early 2010s (so called 'double dip'), which led to a rise of the unemployment rate by nearly 2 pps. over 2011-2013;

(Continued on the next page)

^{(&}lt;sup>1</sup>) The threshold is set such that small shocks remain under the responsibility of the Member States. In addition, the reference period of 15 years is a compromise. If the reference period it too short, recent cyclical developments weigh heavily on the average. A too long reference period may not sufficiently reflect recent reforms and structural changes in the labour market.

^{(&}lt;sup>2</sup>) Based on historical data, for the proposed threshold of 1 percentage point, the frequency at which the conditions would be met is slightly above 10% (i.e. once per decade for a country).

Box (continued)

iii) A 'catastrophic' shock, similar to the one of the late 2000s, which resulted in a cumulated increase in the unemployment rate by around 3 pps. over 2008-2010.

			,					Memo items			
	Latest unemploy ment rate	15-year unemploy ment average	Unemploy ment rate 4 quarters ago	Level condition: ''Unempl. gap''	Change condition: y-o-y change	Double condition	Peak unemploy ment (since	Change from peak	10-year unemploy ment average	Unemploy ment rate in 2007	
	(a)	(b)	(c)	(a)-(b)	(d) =(a)-(c)	(a)>(b) &(d)>1	(e)	(a)-(e)	(f)	(g)	
BE	5.8	7.7	6.1	-1.9	-0.3	NO	8.7	-2.9	7.7	7.5	
DE	3.2	6.5	3.5	-3.3	-0.3	NO	6.2	-3.0	5.1	8.5	
EE	4.6	8.5	5.4	-3.9	-0.8	NO	14.3	-9.7	9.3	4.6	
IE	5.0	9.4	5.9	-4.4	-0.9	NO	16	-11.0	11.3	5.0	
EL	18.5	16.9	21.0	1.6	-2.5	NO	27.7	-9.2	20.8	8.4	
ES	14.1	17.2	16.1	-3.1	-2.0	NO	26.2	-12.1	20.8	8.2	
FR	8.8	9.3	9.2	-0.5	-0.4	NO	10.5	-1.7	9.7	8.0	
IL	10.6	9.4	11.0	1.2	-0.4	NO	12.7	-2.1	10.5	6.1	
CY	7.6	8.9	10.3	-1.3	-2.7	NO	16.6	-9.0	11.1	3.9	
LV	7.0	11.3	8.2	-4.3	-1.2	NO	17	-10.0	12.7	6.1	
LT	6.0	9.8	6.5	-3.8	-0.5	NO	16.8	-10.8	11.2	4.3	
LU	5.3	5.2	5.5	0.1	-0.2	NO	6.5	-1.2	5.5	4.2	
MT	3.5	5.9	3.9	-2.4	-0.4	NO	6.6	-3.1	5.5	6.5	
NL	3.4	5.4	4.1	-2.0	-0.7	NO	7.8	-4.4	5.6	4.2	
AT	4.7	5.2	5.4	-0.5	-0.7	NO	6.2	-1.5	5.3	4.9	
РТ	6.5	11.0	7.6	-4.5	-1.1	NO	17.3	-10.8	12.1	9.1	
SI	4.3	7.1	5.6	-2.8	-1.3	NO	10.7	-6.4	7.9	4.9	
SK	5.8	12.2	7.1	-6.4	-1.3	NO	14.4	-8.6	11.6	11.2	
FI	6.6	8.0	8.0	-1.4	-1.4	NO	9.4	-2.8	8.3	6.9	
EA19	7.8	9.6	8.5	-1.8	-0.7	NO	12.1	-4.3	10.3	7.5	

Table 1: Assessing stabilisation needs via the double condition at the current juncture

Note: The latest unemployment rate refers to 2019Q1, with the exception of EE, EL, IT, CY, LV, AT for which it is 2018Q4 **Source**: Eurostat

Table 2 summarises the main results of the three simulations over 2019-2021. A moderate downturn could result in the double condition being met in a small number of Member States (four in this case) and for a limited period of time. In the event of a more significant downturn, the double condition would be met by a more notable fraction of Member States, representing together more than a third of euro area GDP (though not necessarily all simultaneously). Finally, in the scenario of a severe recession, more than half of euro area countries would meet the double condition at some point. The results are broadly robust to a different parametrisation, i.e. reducing the average unemployment rate from 15 to 10 years.

Overall, the simulations suggest that only a significant deterioration of the economic conditions, and consequently of the labour markets, could give rise to widespread stabilisation needs in the euro area. While current projections do not point in that direction, risks remain tilted to the downside.

		Frequency of activation	Member State (no. of quarters)	% of EA GDP	
	Moderate slowdown (early 2000s)	5.7%	LU(7), NL(3), AT(1), EL(1)	12.1	
	Significant downturn (double-dip)	19.6%	ES(5), NL(4), PT(3), SI(1), AT(1), CY(9), EL(10), IT(8)	36.2	
	Catastrophic shock (late 2000s)	33.0%	ES(9), FI(4), FR(4), IE(8), LT(9), SI(2), AT(3), CY(4), EE(8), EL(8), IT(3), LV(7)	56.7	
re: The frec nber of ob nputed on irce : Europe	uency of activation is measure servations tested (from 2019Q2 to to the maximum number of cou ean Commission	d as the numbe o 2021Q4 for 19 ntries satisfying si	r of times the double co Member States). The perc multaneously the double	ndition is me centage of er condition.	t over the uro area C

Box (continued)

43

Box 2.5: Alternative measure of fiscal sustainability challenges for the fiscal stance analysis

The fiscal stance analysis traditionally uses the Commission S1 fiscal sustainability indicator as a way to measure fiscal sustainability challenges. The S1 indicator captures risks to fiscal sustainability over the medium-term by measuring the cumulated gradual change in the structural primary balance (over 5 years) in order to bring the general government debt to GDP ratio to 60% over the following 15 years. It is a well-established indicator of medium-term fiscal sustainability risks with several strengths in terms of theoretical background (the finite version of the inter-temporal budget constraint), comparability across EU Member States (by relying on commonly agreed methodologies and assumptions), coverage (by accounting for the future spending due to population ageing), and anchor to the reference SGP debt threshold.

While the S1 indicator provides a relevant basis, additional measures of fiscal sustainability challenges can complement the fiscal stance analysis. Here, for robustness purposes, the results of the S1 indicator are complemented by another quantitative metrics. (¹) Such an alternative measure is based on the standard DSA debt threshold (90% of GDP) used by the Commission (see Fiscal Sustainability Report 2018), to identify countries at *high* fiscal sustainability risks (according to this criterion). (²) Furthermore, this alternative measure relies on different interest rate assumptions than the ones traditionally used in the calculation of the standard S1 indicator. In particular, market interest rates are assumed to converge to country-specific values over the medium-term, set in line with financial markets' expectations, rather than a common target (a long-term interest rate of 5% in nominal terms as in the S1; see Box 4.2).





Note: The red, yellow and green circles indicate the high, medium and low sustainability risk level according to the value of the \$1 indicator. **Source:** European Commission

The outcomes of this alternative indicator suggest lower fiscal sustainability challenges in euro area countries, although these results should not be directly interpreted in terms of medium-term risk classification. The lower required adjustment in the structural primary balance as implied by a higher debt target to be reached over the medium-term, as well as more favourable interest rate dynamics imply a reduced required fiscal effort in euro area countries. The difference in terms of cumulative fiscal effort between the two fiscal sustainability measures, namely the S1 indicator and the alternative indicator is around 3.7 pps. of GDP on average for the euro area, and is significant for most countries (see Graph 1). Nevertheless, even with the alternative indicator, a large sustainability gap persists in Italy. These results, provided for illustrative purpose, should not however be directly interpreted in terms of medium-term risk classification.

^{(&}lt;sup>1</sup>) See also the European Commission Report on public finances in EMU 2016, where a detailed comparative analysis of different fiscal sustainability measures is provided.

^{(&}lt;sup>2</sup>) The DSA risk classification is based on a wide range of indicators, criteria and scenarios (e.g. in addition to the debt level, the debt path, and the plausibility of fiscal assumptions are considered under different scenarios). Stochastic projections also contribute to the risk classification.

Box 2.6: Are SCPs moving toward growth-friendlier composition of expenditure and revenue? A tentative score-based assessment

Relevance of growth-friendlier composition of expenditure and revenue

In a context of higher public indebtedness following the fiscal crisis, the need for improving the quality of both spending and revenue has become more prominent in the political agenda of national government. Empirical findings support the idea that sound and sustainable fiscal policies are preconditions for growth. When governments accumulate large debt, long-run growth could be hampered. Beyond the fiscal aggregates, acting on the composition of fiscal policy represents a powerful policy lever, which can, in principle, be pulled even if a country needs to reduce its deficit. Indeed, tax instruments as well as expenditure items have different effects on growth (¹).

A simple score to assess the quality of budget composition

Following Cournède et al, (2014), we compute a simple indicator to assess the change in the growthfriendliness composition of the budget between 2019 and 2022 based on the budgetary composition planned in the Stability and Convergence Programmes. (²) Two scores are calculated, the first indicates the quality of the spending/revenue composition relative to short-term economic growth, the second relative to long-term growth. The composition score is a weighted average of the OECD score of individual spending/revenue items based on the share of this item in GDP.

Composition score
$$_{jt} = \sum_{i=1}^{n} GDP$$
 share $_{jit} * growth \ score_{it}$

With *j* referring to the specific country, *t* to the year for which the assessment is conducted, *i* to the individual budget item, and the growth score expressing, in numerical terms (from -1 to +1), the strength of growth effect as displayed in Table 1. Therefore, for each country what matters is not only the composition of expenditure and revenues but also the size of the government, and in particular the tax burden.

The composition scores, although intuitive, should only provide a first assessment, given the numerous caveats of such approach. The score only considers the composition of public finance without any reference to microeconomic efficiency of individual items. The computation of the scores is also limited by the availability of relevant data in the SCPs, in particular the limited level of disaggregation on the expenditure side (see footnote 2). Lastly, it is affected by the standard limitation of indicator-based approach, which invites for a more in-depth analysis.

The fiscal items, used for the analysis, cover most of expenditure (84%) and revenue (90%) (Graph 1). One third of the expenditure corresponds to social transfers (pension, unemployment insurance, family allowance, sickness/disability). Another third covers government consumption other than those dedicated to education and health. One sixth of total spending corresponds to "productive spending", covering: public investment, education and health. The last sixth represents subsidies and debt servicing. In the scoring, we only consider the primary spending, at the exclusion of interest payments, which are not directly under government control (albeit a sound debt management may help). Revenue is almost equally divided across the three main categories, namely direct taxes, indirect taxes and social security contributions. The weights

^{(&}lt;sup>1</sup>) See for instance A. (2008). "The quality of public finances and economic growth". *Economic Papers 337*, European Commission; Barrios S. and Schaecter, A. (2009). "Gauging by numbers: A first attempt to measure the quality of public finances in the EU". *Economic Papers 382*, European Commission; Cournède, B., Goujard A. and Pina Á. (2014). "Reconciling fiscal consolidation with growth and equity". *OECD Journal: Economic Studies*, Vol. 2013/1.

^{(&}lt;sup>2</sup>) The series of Education, Health services and Other government consumption are determined on the base of their percentages in terms of Final consumption expenditure, as they appear in COFOG decomposition for 2017. Seemingly, the series for Pensions, Sickness and disability payments and Family are reconstructed on the base of their percentages in terms of Social transfer other than in kind.

Box (continued)

used for the scoring are expressed as a percentage of GDP, to also take into consideration their economic size.

Table 1: Growth friendliness of fiscal instruments ("OECD scores")

Growth assessment Short-term Long-term Expenditure side Education Health services ++ +Other government consumption ++ Pensions Sickness and disability payments + Unemployment benefits +Family + ---Subsidies 4 Public investment ++ ++ Revenue side Current taxes on income, wealth, etc Capital taxes ___ Social security contributions Taxes on production and imports Property income

Note: As illustrated in Table 1, the growth effect assumes four levels of intensity "--", "-", "0", "+" and "++", which are translated in numerical terms in a scale from -1 ("- -") to 1 ("+ +"), with consecutive intensities differentiating by 0.5. Source: Elaborations on Cournède et al, (2014)

Graph 1: 2019 EU average weights of individual fiscal item (% of total expenditure and revenues respectively)



Source: European Commission based on 2019 SCPs

Overall, the SCPs do not point to an improvement in the quality of budget composition by 2022

Regarding the impact on short-term growth (Graph 2a), most countries move slightly toward a less growthfriendly composition of the budget from 2019 to 2022. Regarding the impact on long-term growth (Graph 2b), instead, no significant change is observable in the quality of budget composition. However, this could also be due to unavailability of disaggregated consumption and social expenditure. Member States - in particular those with a low composition score in 2019 - should consider stepping up effort to reduce unproductive spending (including interest payment by pursuing prudent fiscal policies), to boost investment and other growth-friendly spending while reducing the share of most distortionary taxes.





Note: light blue bars in panel b) represent 2019 SCP projections **Source:** European Commission and 2019 SCPs

Stability Programmes project a steady increase in private and public investment in the 2019-2022 programme period. Projections put forward in the 2019 Stability Programmes would result in an overall increase in the euro area investment ratio of 1.2 pps. of GDP by 2022. The increase in the public investment ratio would be 0.3 pps. of GDP (Graph 1b), while the private investment ratio is set to increase by 0.9 pps. of GDP. At the same time, a projected increase in the saving ratio close to 1 pp. of GDP by 2022 can be inferred given the decline in the surplus with the rest of the world (0.3 pps. of GDP).

There seems to be scope to increase public investment in the euro area, especially in Member States that have not accumulated public debt for other purposes. Graph 2 presents the relationship between the existing stock of net debt and the capital stock of the general government. The main intuition behind the graph is that when net debt is lower than the capital stock, it could be considered that the accumulation of debt has been fully related to finance public investment. Conversely, when net debt is higher than the capital stock, the exceeding part of net debt has not been used to finance public investment. Member States that need to enhance their public infrastructure and do not have sustainability challenges may prioritise investment spending to debt reduction taking advantage of the current low level of interest rates. This might be an appropriate policy also from an intergenerational point of view in particular if the accumulation of debt in the past has been mainly related to the accumulation of public capital, which will bring benefit to future generations. Scaling-up of public investment in the euro area should be done without endangering fiscal sustainability. For Member States with limited fiscal space and where the accumulation of high public debt has not been reflected in higher capital stock, this would require improvements in the composition of public finances to create room for more investment.

(Continued on the next page)



Note: Net public debt is computed as gross debt (Madistinchi definition) her of financial assets hera by the GG sector in currencies and deposits, loans and debt securities. GG capital stock data (in current price replacement costs) come from Eurostat national accounts balance sheets by sector for non-financial assets [nama_10_nfa_bs], with the following exceptions: data for ES, IE and MT are European Commission – DG ECFIN estimates, which are derived from Eurostat general government GFCF and consumption of fixed capital and AMECO data for net capital stock for the total economy. Moreover, 2017 values for CY, EE, EL, LT and LV are estimated by adding to 2016 capital stock data general government GFCF in 2017, and then subtracting consumption of fixed capital in 2017. **Source:** European Commission

3. RISK ASSESSMENT OF MEMBER STATES' PLANS

This section analyses possible risks to the achievement of the budgetary targets set out in the 2019 Stability and Convergence Programmes (SCPs). To ensure the comparability with Commission 2019 spring forecast (²⁸), the assessment of risks is conducted separately and differently over the two sub-periods (2019-2020 and 2021-2022), composing the SCPs horizon. The section is structured as follows. Section 3.1 presents a comparison between the main fiscal variables for 2019 and 2020, as planned in the SCPs and as forecast by the Commission. Section 3.2 checks the feasibility and internal consistency of Member States' budgetary targets for 2021 and 2022.

3.1. RISKS TO 2019 AND 2020 PLANS

The difference in projections between Commission and Member States identifies the risks while their seriousness depends on the grade of optimism of the Member States plans. Table 3.1 presents the main macroeconomic and fiscal variables considered for the risk assessment. In most Member States, the programmes are based on independently produced or endorsed macroeconomic forecasts (Box 3.1 for details).

Table 3.1: G	e 3.1: GDP growth and main fiscal variables: a comparison between 2019 SCPs and Commission forecast										
		2018	20)19	2	020					
		Outcome (COM)	Commission forecast	2019 SCP	Commission forecast	2019 SCP					
EU	Real GDP growth	2.1	1.5	1.5	1.7	1.7					
euro area	(%)	1.9	1.2	1.4	1.5	1.6					
EU	GDP deflator	1.6	1.8	1.8	1.8	1.8					
euro area	(% change)	1.4	1.6	1.6	1.6	1.7					
EU	Headline balance	-0.6	-1.0	-0.9	-1.0	-0.6					
euro area	(% GDP)	-0.5	-0.9	-0.9	-0.9	-0.5					
EU	Interest expenditure	1.8	1.8	1.7	1.7	1.7					
euro area	(% GDP)	1.8	1.8	1.7	1.7	1.6					
EU	Change in the structural balance	0.3	-0.2	-0.2	-0.2	0.2					
euro area	(% potential GDP)	0.3	-0.2	-0.2	-0.3	0.1					
EU	Fiscal effort based on	-0.2	-0.6	-0.3	-0.4	0.0					
euro area	expenditure benchmark (% GDP)	-0.3	-0.5	-0.4	-0.4	0.0					

Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

The growth and inflation outlook for 2019 and 2020 are similar based on the Commission forecast and the aggregate SCP plans. The Commission projects slightly lower real GDP growth for the euro area in both 2019 and 2020, while projections for the EU are the same. As a result, the impact of these differences on the aggregate headline balance is expected to be marginal.

The SCP budgetary targets for the 2019 headline budget balance (*HB*) are very close to the Commission forecast in aggregate terms. Hence, the risks relative to the achievement of 2019 fiscal targets seem to be limited at aggregate level. At country level, only four countries (BE, EL, RO, and SK) plan 2019 headline balance budgets higher than 0.3 pps. of GDP compared with the Commission forecast (Graph 3.1). By contrast, the Commission forecast is more favourable by at least 0.3 pps. of GDP in the case of four countries (BG, DK, HR and LU).

^{(&}lt;sup>28</sup>) Commission 2019 spring forecast covers 2019 and, based on unchanged policy, 2020. Thus it provides a benchmark against which to assess the macroeconomic and budgetary projections only for the sub-period 2019-2020 of the SCP horizon.

Box 3.1: Independent production or endorsement of the macroeconomic forecasts underpinning the 2019 Stability and Convergence Programmes

Acknowledging the crucial role that realistic macroeconomic forecasts play in budgetary processes, Article 4(4) of the Two-Pack Regulation (EU) No 473/2013 lays down, inter alia, the requirement that euro area Member States shall base their national medium-term fiscal plans on independent macroeconomic forecasts. Article 2(1b) of the Regulation defines independent macroeconomic forecasts as "macroeconomic forecasts produced or endorsed by independent bodies".

Out of the 19 Stability Programmes submitted in spring 2019, five (AT, BE, LU, NL and SI) relied on macroeconomic forecasts prepared by external independent producers. In addition, a special mechanism is in place in FI, ensuring that the macroeconomic forecast is produced independently within the Ministry of Finance by the Economics Department, which is operationally independent from the Budget Department.

In the other 13 Member States (CY, DE, EE, EL, ES, IE, IT, FR, LV, LT, MT, PT and SK), the official macroeconomic scenario was produced by the government and subsequently endorsed by an independent fiscal institution. In two countries, the relevant institutions are not the national fiscal councils (i.e. independent entities in charge of monitoring public finances and compliance with numerical rules), but specialised committees tasked with assessing the government forecasts. More precisely, in SK, the endorsement competence was conferred to the Macroeconomic Forecasting Committee, which is composed of representatives of independent institutions (commercial banks, the Slovak Academy of Sciences and the central bank). In DE, the Joint Economic Forecast Project Team, comprising leading economic research institutions, was established in 2018 to act as the independent body for endorsing the federal government's macroeconomic scenario.

As regards the Stability Programmes of the 13 Member States in which independent bodies endorse the forecasts, in most of the cases those bodies assessed the official macroeconomic scenario as prudent, and therefore, at most, pointed to broadly-balanced risks to the projections. In a few countries, most notably EL and IT, the endorsement texts highlighted downside risks, of both internal and external nature. For PT, the Public Finance Council endorsed the macroeconomic forecasts underlying the Stability Programme for the period 2019-2020, but not for the period 2021-2023, for which it flagged significant divergences vis-à-vis other institutional projections; overall, the government's forecast for the outer years of the programme was considered to be neither the most likely, nor the most prudent scenario.

Outside the euro area, there is no corresponding legal requirement; this being said, the UK uses the macroeconomic forecasts produced by the independent Office for Budgetary Responsibility for both annual and medium-term budgetary planning since its establishment in 2010. Furthermore, a systematic scrutiny has been introduced in CZ with the establishment of the Committee on Budgetary Forecasts, composed of leading economists from the private, public and academic sectors. The Committee is tasked with assessing the plausibility of the government's macroeconomic and budgetary forecasts. In other non-euro area Member States, even though independent fiscal institutions typically pronounce on the realism of macroeconomic projections (more often for annual budgets than for the multi-year fiscal plans), their opinions have no institutional bearing in the policy-making process.

In order to detect potential drivers of the differences between the two projections, further analysis is performed. The decomposition of the difference is based on three potential drivers described as follow:

$$HB_{t}^{COM} - HB_{t}^{SCP} = (HB_{t-1}^{COM} - HB_{t-1}^{SCP}) + (nom. growth_{t}^{COM} - nom. growth_{t}^{SCP}) * \varepsilon + Residual$$

Base effect Growth gap Policy gap

 $HB_t^{COM} - HB_t^{SCP}$ represents the difference in the headline budget balance in 2019 according to the two projections. $HB_{t-1}^{COM} - HB_{t-1}^{SCP}$ is the *Base effect*, reflecting possible differences in the 2018 balance budget outcome. nom. growth $t_t^{COM} - nom. growth t_t^{SCP}$ multiplied by ε (the semi-elasticity) is the growth gap (²⁹), and then there is a residual. This last component, defined as policy gap, among others, measures the difference in the assessment of budgetary measures underlying the projected fiscal targets. It can also include possible differences in revenue elasticities or other unexplained factors beyond the control of the government, such as for instance assumptions about interest rates. A negative residual represents a downside risk to the plans. Graph 3.1 illustrates that for most Member States the main driver of differences between the Commission budget balance and the SCP targets for 2019 is related to the policy gap.



Note: The graph shows a decomposition of the difference between the balance budget figure in 2019, as per the SCPs and Commission forecast into (i) base effect; (ii) difference in a standardized measure of the growth gap impact; and (iii) a residual. How to read the graph: negative (positive) values imply that the component has a positive (negative) impact in the SCPs relative to the Commission 2019 spring forecast. The sum of the components is the difference between the COM headline budget balance and the SCP headline balance forecast.

SCP budgetary targets are more favourable than the Commission forecast for 2020 at aggregate level. For both the EU and the euro area, the aggregate headline deficit is lower based on the SCPs budgetary targets than according to the Commission no-policy-change forecast, by 0.4 pps. of GDP (Graph 3.2). The difference is mainly driven by the *policy gap*, while the different starting point (*base effect*) and the difference due to macroeconomic assumptions (*growth gap*) are very limited in aggregate terms. For the euro area aggregate, the Commission forecast expects lower revenues and higher primary expenditure than the Stability Programmes in 2020. The latter might also be explained by the lower growth assumptions (Table 3.1).

^{(&}lt;sup>29</sup>) Different nominal growth forecast for a given year can result in different forecast for the headline balance. The standard semielasticities are used to approximate the effect of different nominal growth assumptions on headline balance prospects. For more information on semi-elasticity computations, see Mourre G., Poissonnier, A. and Lausegger, Martin (2019). "The Semi-Elasticities Underlying the Cyclically-Adjusted Budget Balance: An Update and Further Analysis", *Discussion Paper 098*.



Note: The graph shows a decomposition of the difference between the balance budget figure in 2020, as per the SCPs and Commission forecast into (i) base effect; (ii) difference in a standardized measure of the growth gap impact; and (iii) a residual. How to read the graph: negative (positive) values imply that the component has a positive (negative) impact in the SCPs relative to the Commission 2019 spring forecast. The sum of the components is the difference between the COM headline budget balance and the SCP headline balance forecast.

At country level, the *policy gap* explains the large difference between the Commission projections and SCP targets for 2020 in five Member States. Graph 3.2 shows that for nine Member States (SE, PT, SK, ES, EL, BE, IT, PL and RO) the Commission forecast projects a 2020 budget balance lower the SCP targets by at least 0.3 pps. of GDP. The *policy gap* is particularly big in the case of ES, BE, IT, PL and RO (also EL if 2019 is considered), which implies an overall difference close or above 1 pp. of GDP. (³⁰) A large policy gap is a recurrent feature of the risks analysis of the SCPs' targets because the Commission forecast is made under an unchanged policy assumption. This implies that it only considers measures that are deemed sufficiently well specified and credibly announced. More favourable growth projections for 2020 in the SCPs than in the Commission forecast also have a sizeable impact (at least 0.3 pps. of GDP) in number of Member States (MT, HU, LV, LU and IT).

The change in the structural balance planned in the SCPs for 2020 is 0.4% of GDP higher than the Commission forecast. SCPs were in many cases only available after the cut-off date for the Commission 2019 spring forecast or contain targets that are not underpinned by measures already adopted.

3.2. RISKS TO 2021 AND 2022 PLANS

On the revenue side, risks to the achievement of the outer years' fiscal targets can stem from the plausibility of budgetary projections at unchanged policy, and from the impact of fiscal measures. Contrary to the assessment of risks to the fiscal targets for 2019 and 2020 – where Commission forecasts provide a natural benchmark against which to assess SCPs' projections – the evaluation of risks in the

^{(&}lt;sup>30</sup>) In the Italian case, the difference reflects the no-policy change scenario underlying the SCP which includes already legislated VAT hikes (around 1.3% of GDP) that are not included in the Commission forecast.

outer years of the programme mainly focuses on the comparison between the fiscal targets and no-policy change projections in the programmes. The comparison of these two sets of variables provides a direct element of analysis to gauge the magnitude of the first source of risks: the cumulative amount of additional measures Member States would need to implement according to their own projections in order to reach their deficit objectives. While all Member States submitted no-policy change revenue projections, only twenty-one of them submitted no-policy change expenditure ones. (³¹) Therefore, this section focuses more specifically on the revenue-side.

							_		Annual		
	Tar	otod		of wh	ich:		Repo	orted	reve	enue	
	revenu	ige in le ratio	Chan revenu at uncl pol	ge in e ratio nanged icy	Imp anr meas	lied wal sures	Discret Reve Meas	tionary enue sures	windfa shorti implici 2018	alls (+) / falls (-) t in the S SCPs	
		(1)	(1	I)	()=	(I)-(II)	(ľ	V)	(V)	
pps. of GDP	2021	2022	2021	2022	2021	2022	2021	2022	2021	2022	
BE	0.0	0.0	-0.2	0.1	0.2	-0.1	0.2	-0.1	-0.2	0.1	
CY	-0.3	-0.5	-0.4	-0.5	0.2	0.0	0.3	0.1	-0.4	-0.5	
DE	-0.3	0.0	-0.3	0.0	0.0	0.0	-0.4	-0.1	-0.3	0.0	
EE	-0.7	-1.0	-0.7	-1.0	0.0	0.0	0.2	-0.4	-0.7	-1.0	
EL	-0.8	-0.9	-0.9	-0.8	0.1	-0.1	-0.1	-0.1	-0.9	-0.8	
IE	-0.1	0.0	-0.1	0.0	0.0	0.0	0.0	0.0	-0.1	0.0	
ES	0.5	0.4	0.4	0.4	0.0	0.0	0.1	0.0	0.4	0.4	
FR	-0.3	-0.3	0.0	-0.1	-0.3	-0.2	-0.3	-0.3	0.0	-0.1	
п	-0.1	-0.4	-0.1	-0.3	0.0	-0.1	0.1	-0.1	-0.1	-0.3	
LV	-0.5	-0.5	-0.5	-0.5	0.0	0.0	0.0	0.0	-0.5	-0.5	
LT	-0.1	-0.3	0.3	-0.3	-0.4	0.0	0.0	0.0	0.3	-0.3	
LU	0.1	0.0	0.1	0.0	0.0	0.0	0.0	0.0	0.1	0.0	
MT	-1.0	-0.5	-1.0	-0.6	0.0	0.0	0.0	0.0	-1.0	-0.6	
NL	-0.6	-0.1	-0.6	-0.1	0.0	0.0	-0.4	0.2	-0.6	-0.1	
AT	-0.6	-0.5	-0.1	-0.1	-0.5	-0.4	-0.5	-0.4	-0.1	-0.1	
РТ	0.3	-0.7	0.3	-0.7	0.0	0.0	0.4	-0.4	0.3	-0.7	
SI	-0.8	-0.5	-0.8	-0.5	0.0	0.0	0.0	0.0	-0.8	-0.5	
SK	-0.5	-0.2	-0.4	-0.2	-0.1	0.0	-0.2	0.0	-0.4	-0.2	
FI	-0.3	-0.1	-0.3	-0.1	0.0	0.0	-0.2	0.0	-0.3	-0.1	
BG	-0.4	-0.8	-0.4	-0.4	0.0	-0.4	0.4	0.0	-0.4	-0.4	
cz	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	
DK	0.0	0.2	0.1	0.2	-0.1	0.0	-0.1	0.0	0.1	0.2	
HR	-0.5	-0.2	-0.5	-0.2	0.0	0.0	0.0	0.0	-0.5	-0.2	
HU	-1.4	-1.7	-1.4	-1.7	0.0	0.0	-0.4	-0.6	-1.4	-1.7	
RO	-0.3	0.4	-0.3	0.4	0.0	0.0	0.0	0.0	-0.3	0.4	
PL	-1.2	-1.1	-1.2	-1.0	0.0	-0.1	0.2	0.1	-1.2	-1.0	
SE	-0.1	-0.2	-0.1	-0.2	0.0	0.0	-0.1	0.0	-0.1	-0.2	
UK	-0.2	0.3	0.0	0.3	-0.2	0.0	0.0	0.0	0.0	0.3	
EU	-0.3	-0.1	-0.2	-0.1	-0.1	-0.1	-0.1	-0.1	-0.2	-0.1	
FΔ	-(12	-0.2	-0.2	-0.1	-01	-0.1	-0.2	-(11	-0.1	-0.1	

 Table 3.2:
 Implicit amount of revenue measures and revenue windfalls for 2021 and 2022 (% of GDP)

Note: Windfalls/shortfalls in column (v) are approximated by comparing the no-policy-change revenue projections in the SCPs with a simplified calculation based on a revenue elasticity of one with nominal GDP growth projections. **Source:** European Commission based on 2019 SCPs

Overall, the revenue projections at unchanged policy seem to be prudent. Most Member States include some revenue shortfalls over 2021-2022 in their unchanged policy projections, as shown in Table 3.2, column (v).(32) Only three Member States, count on substantial revenue windfalls in their unchanged policy scenario, ranging from 0.3% of GDP (DK, UK) to 0.8% of GDP (ES). If these windfalls were not

^{(&}lt;sup>31</sup>) The submission of no-policy change revenue projections has been agreed upon by all Member States according to the code of conduct of the Stability and Growth Pact. Conversely, the submission of no-policy change expenditure projections remains voluntary. <u>http://ec.europa.eu/economy_finance/economic_governance/sgp/pdf/coc/code_of_conduct_en.pdf</u>.

^{(&}lt;sup>32</sup>) The figures in this table should however be interpreted with care, as some Member States might have included the impact measures already decided in their no-policy-change revenue projections. In such a case, apparent revenue shortfalls might in fact result from tax cuts already decided while revenue windfalls might result from tax increases that have already been decided. However, this distortion is likely to be limited for the outer years of the programmes.

to materialise, these Member States would need to implement larger revenue measures than reported to attain their revenue targets in 2021-2022.

Risks stemming from the size of the required measures seem restrained on the aggregate. The difference between the unchanged policy projections and the revenue targets indicate the amount of revenue measure that needs to be taken to implement the programme. On aggregate, the amount of discretionary measures required in either 2021 or 2022 (Table 3.2, column iii) is consistent with some revenue-decreasing measures (i.e. tax cuts). At country level, CY will need to implement additional revenue measures of around 0.2% of GDP in 2021-2022, but the programme plan an even large amount of discretionary revenue measures (0.4% of GDP).

On aggregate, interest expenditure (Table 3.3) it is not expected to drop further in 2021-2022. The implicit interest rate on the outstanding debt stock is expected to slightly increase for the EU between 2020 and 2022, reaching 2.1% while remain stable for the euro area (2.2%). At country level, 20 countries still count on savings from lower interest expenditure in the outer years of the programme. These are significant in particular for CY, PT, SI, HR and especially LT. On the other hand, FR, IT, EE and FI expect an increase in 2021-2022. A comparison with the Commission's own medium term debt projections shows that interest rate assumptions of the SCPs appear on aggregate relatively prudent. However, they appear substantially optimistic in the case of LT, LU, SK, DK, RO and SE. This being said, the evolution of the implicit interest rate depends on a number of assumptions, for example regarding the future maturity structure, and therefore comparisons between projections are only indicative.

		2019	2020	2021	2022
			2019	SCP	
EU	Real GDP growth	1.5	1.7	1.6	1.6
euro area	(%)	1.4	1.6	1.4	1.3
EU	GDP deflator	1.8	1.8	1.8	1.8
euro area	(% change)	1.6	1.7	1.7	1.7
EU	Headline balance	-0.9	-0.6	-0.5	-0.3
euro area	(% GDP)	-0.9	-0.5	-0.3	-0.2
EU	Interest expenditure	1.7	1.7	1.6	1.7
euro area	(% GDP)	1.7	1.6	1.6	1.7
EU	Change in the	-0.2	0.2	0.1	0.2
euro area	structural balance (% potential GDP)	-0.2	0.1	0.1	0.1
EU	Fiscal effort based on	-0.3	0.0	0.0	0.1
euro area	(% GDP)	-0.4	0.0	0.0	0.1

4. SUSTAINABILITY

The sustainability of public finances across Member States, against the background of revised macroeconomic scenario, fiscal outlook, fiscal plans and the demographic ageing is analysed in this section. The analysis presented here takes as a point of departure the latest Commission 2019 spring forecasts and the 2019 Stability and Convergence Programmes (SCPs). The long-term budgetary projections released with the 2018 Ageing Report have also been incorporated in the simulations. (³³) The analysis is notably based on the Commission multidimensional framework to assess fiscal sustainability risks (see Box 4.1). This section also includes a specific sensitivity analysis, considering an alternative assumption for the evolution of interest rates in the medium term, anchored to financial markets' expectations (see Box 4.2).

4.1. SHORT-TERM FISCAL SUSTAINABILITY RISKS

Overall, short-term fiscal sustainability risks have declined for EU countries since 2009, although risks appear on the rise in some cases. The S0 indicator, an 'early-detection indicator', allows identifying short-term fiscal sustainability risks, on the basis of a large set of structural variables. (³⁴) In 2009, more than half of the Member States had values of the S0 indicator above its critical threshold, signalling risks of fiscal stress in the upcoming year. This round, only Cyprus has a value of S0 that is above the threshold signalling an overall fiscal sustainability risks in the short-term, while significant vulnerabilities on the fiscal side are identified in four additional countries (Spain, France Italy and Hungary, see Graph 4.1). These vulnerabilities deserve particular attention, in a context where financial market sentiments can change rapidly. Italy is particularly exposed to sudden changes in such financial market perceptions, notably in the light of its sizeable government financing needs.

In the case of Cyprus, both the fiscal and financial-competitiveness sides of the economy pose potential challenges. In particular, one-off banking support measures in 2018 led to the deterioration of several fiscal variables, such as the primary balance and the net and gross government debt, while macro-financial vulnerabilities remain significant. Nevertheless, in 2019, the general government balance is expected to be in high surplus contributing to a reduction of gross financing needs this year (also considering the high share of concessionary debt), thus overall mitigating the high short-term sustainability risk.

In Spain, France Italy and Hungary, the high level of government debt, coupled in some cases with important gross financing needs (Spain, Italy and Hungary), a weak fiscal position (Spain, France and Hungary) and the weight of short-term government debt as a share of GDP (Italy), contribute to the assessment. Yet, for these four countries, the overall S0 indicator does not point to overall short-term fiscal sustainability risks, in light of more limited macroeconomic imbalances.

^{(&}lt;sup>33</sup>) European Commission (DG ECFIN) and Economic Policy Committee (AWG) (2018), "The 2018 Ageing Report: Economic and budgetary projections for the EU Member States (2016-2070)", *European Economy, Institutional Paper*, No.079.

^{(&}lt;sup>34</sup>) These variables include fiscal and financial-competitiveness variables (see also Box 4.1 for more explanations on this indicator).

Box 4.1: The Commission approach to assess fiscal sustainability

The Commission uses a multidimensional approach to assessing fiscal sustainability. It integrates the longer term with an assessment of more immediate challenges and risks, underpinned with appropriate indicators which can point to the scale and the scope of the sustainability challenges. This multidimensional approach enables assessing:

- Short-term fiscal challenges, through a combination of fiscal, financial and competitiveness indicators aiming at an 'early detection of fiscal stress'. The S0 indicator is an 'early-detection indicator', designed to highlight shorter-term (one-year horizon) risks of fiscal stress stemming from the fiscal, as well as the financial and competitiveness sides of the economy. A set of 25 fiscal and financial-competitiveness variables are used in the composite indicator S0. (¹)If section 4 focuses mainly on medium- and long-term fiscal challenges, a first sub-section discusses these short-term fiscal challenges that are present in a few Member States.
- *Medium-term fiscal challenges*, by looking at the risks to fiscal sustainability over the medium run, based on debt sustainability analysis (DSA) and the S1 sustainability indicator, in line with the Fiscal Sustainability Report 2018. ⁽²⁾The S1 indicator shows the fiscal gap related to the excess of projected government expenditure, including projected age-related expenditure, over projected revenue together with any gap with respect to the steady adjustment in the structural primary balance over the five years after the period covered by the forecast, to bring the debt-to-GDP ratio to 60% of GDP by 2033.
- *Long-term fiscal challenges*, based on the joint use of the S2 sustainability indicator and the overall results of the DSA. The S2 indicator shows the fiscal gap related to the excess of projected government expenditure, including projected age-related expenditure over projected revenue together with any gap with respect to the primary balance needed to ensure that the debt-to-GDP ratio is not on an ever-increasing path over an infinite horizon. Introduced with the Fiscal Sustainability Report 2018, the use of DSA results to assess overall long-term fiscal sustainability risks aims at prudently capturing risks linked to medium to high debt-to-GDP ratios (see Box 4.1 in the FSR 2018 for more explanations).

In addition to these key indicators, the Commission fiscal sustainability framework encompasses additional mitigating and aggravating risk factors that are analysed in its regular Fiscal Sustainability Report and Debt Sustainability Monitor (see for example chapter 5 of the FSR 2018).

⁽¹⁾ The methodology for the S0 indicator is fundamentally different from the S1 and S2 indicators, which quantify the required fiscal adjustment, the 'fiscal gap'. S0 does not assess 'fiscal gaps' but is a composite indicator estimating risks of 'fiscal stress' in the short term, on the basis of risk thresholds (based on the observation of past episodes of 'fiscal stress' for relevant structural variables and their combinations).

^{(&}lt;sup>2</sup>) For details about the sustainability risk classification and the methodology behind the Debt Sustainability Analysis (DSA), see European Commission (2019), 'Fiscal Sustainability Report 2018', European Economy, Institutional papers, No 94, EC, Brussels.



4.2. MEDIUM- TO LONG-TERM FISCAL SUSTAINABILITY: GOVERNMENT DEBT PROJECTIONS

Turning to medium- and longer-term implications for fiscal sustainability, and taking account of the projected changes in age-related expenditure, the macroeconomic scenario and the fiscal outlook and plans, two main scenarios are considered:

- The **COM no-fiscal policy change scenario**, with the structural primary balance to GDP ratio kept constant at 2020 forecasted level as in the Commission 2019 spring forecast (reflecting a "no-fiscal policy change" assumption) (net of future **changes in ageing-related expenditure**).
- The **SCP** scenario reflecting planned changes in fiscal policies as reported in the SCPs (fully in line with SCPs until the end of the programme horizon). Beyond the programme horizon, the structural primary balance as a share of GDP is kept constant (net of future changes in ageing-related expenditure).

Graph 4.2 shows the projected evolution for the government gross debt ratio (including the projected change in ageing-related expenditure), for the EU as a whole. The solid thick line shows the outcome for the COM no-fiscal policy change scenario under the assumption of no fiscal consolidation measures beyond those contained in the Commission 2019 spring forecast (structural primary balance to GDP ratio kept constant at forecasted 2020 level) and incorporates expected future ageing-related spending, as projected in the 2018 Ageing Report. (³⁵)

Government debt is expected to decrease until 2020 and beyond. According to the Commission 2019 spring forecast, government debt will continue to decrease and reach around 79% of GDP in 2020 in the EU as a whole. Given the sustained fiscal surplus until 2020 and the still negative snowball effects until the mid-2020s, debt is projected to continue to decline in the following years. Moreover, the cost of ageing as a share of GDP is projected to rise only slowly in the years to the mid-2020s. By the end of the projection horizon (2029), debt would reach less than 76% of GDP.

The debt path for the EU in the SCP scenario lies well below the path projected in the COM nopolicy-change scenario (a difference of around 9 pps. of GDP between debt ratios in 2029). Indeed,

^{(&}lt;sup>35</sup>) This consists of projections of pension, health care, long-term care, education and unemployment benefit spending. In addition, the projected changes in property income and in taxes on pensions are incorporated.

the SCP scenario would lead to a more marked reduction in the government debt-to-GDP ratio with debt falling to around 67% of GDP by 2029 (dashed line).



Note: The medium-term projections are based on the Commission 2019 Spring forecast (up to 2020) and on the 2019 SCPs, the updated t+10 projections and the projections in the 2018 Ageing Report. The output gap is assumed to close in t+5. The inflation rate (GDP deflator) is assumed to converge linearly to 2% in t+5, when the output gap is closed and remains constant thereafter, for all countries. The long-term interest rate on new and rolled over debt is assumed to converge to 5% (in nominal terms) by the end of the 10-year projection horizon, based on the AWG-EPC agreed assumption, while the short-term interest rate on new and rolled over (0.83). The structural primary balance is kept unchanged after either the end forecast or the end programme year, apart from the projected change in age-related expenditure according to the AWG reference scenario from the 2018 Ageing Report. The primary balance is adjusted by using the budget sensitivities in the period until the output gap is assumed to be closed in t+5. No stock-flow adjustment is assumed after the end of forecast or programme horizon.

4.3. MEDIUM-TERM FISCAL SUSTAINABILITY CHALLENGES

Developments in the Commission no-policy-change scenario

The adjustment needed in the medium-term with respect to unchanged policies is calculated as the additional fiscal adjustment required up to five years ahead, in order to reach a debt-to-GDP ratio at 60% by 2033 (see Graph 4.3). (³⁶) The improvement relative to the COM no-policy-change scenario required in the structural primary balance to achieve a debt-to-GDP ratio target of 60% by 2033 amounts to 1.9 pps. of GDP over the period 2021–2025 in the EU as a whole, i.e., an average annual fiscal consolidation effort of around 0.4 pp. per year. In other words, the structural primary balance in the EU should improve from a forecasted surplus of 0.5% of GDP in 2020 (structural balance of -1.3% in 2020) to a surplus of close to 2.4% in 2025.

According to the S1 indicator, 5 countries face *high* risk (BE, ES, FR, IT and PT), 4 countries face *medium* risk (HU, PL, RO and UK), and 18 countries face *low* risk over the medium-term. For the majority of Member States, the overall medium-term risk classification coincides with classification based on the S1 indicator. However, for some countries (HR, CY, RO and UK) it worsens with the debt sustainability analysis pointing to additional risks (see Table 4.1). (³⁷) Considering an alternative

^{(&}lt;sup>36</sup>) The base year is either the end of the forecast horizon (2020) or the end of the SCP horizon (country specific, with values between 2022 and 2023), depending on the scenario considered.

^{(&}lt;sup>37</sup>) For details about the sustainability risk classification and the methodology behind the Debt Sustainability Analysis (DSA), see European Commission (2019), 'Fiscal Sustainability Report 2018', *European Economy, Institutional papers*, No 94, EC, Brussels.

assumption for interest rates (based on financial markets' expectations, see Box 4.2) would drive down the value of the S1 indicator in almost all Member States, but would not change the medium-term risk classification.

Developments assuming implementation of the fiscal plans in the SCPs

If the fiscal plans in the SCPs were fully implemented and, additionally, not weakened after the end of the programme horizon, the fiscal gap would be reduced by around one half (0.8% of GDP instead of 1.9%) in the no-policy-change scenario. (³⁸) Consequently, risks would be reduced and according to the S1 indicator, 5 countries would face *high* risk, 1 country would face *medium* risk and 21 countries would face *low* risk. In the SCP scenario, 3 Member States would go to a lower risk category, respectively HU, PL and RO, moving from *medium* to *low* risk. Only in the cases of NL, BG and DK, the S1 indicator based on the SCPs would be higher than the one based on the Commission no-policy-change scenario, mainly as the result of the assumed lower structural primary surplus. This would not lead, however, to an unfavourable change in the risk category in any of the three countries.



4.4. LONG-TERM FISCAL CHALLENGES

Developments in the Commission no-policy-change scenario

In the long term, the sustainability of the fiscal position is assessed by the gap relative to the primary balance required to stabilize debt and cover all the future changes in age-related expenditures. Overall, the EU as a whole has a long-term fiscal sustainability gap of 2.3 pps. of GDP. The cost of ageing (CoA), based on the 2018 Ageing Report projections, contributes by 1.7 pps. of GDP to the gap, and the initial budgetary position (IBP) by 0.5 pps. of GDP. Graph 4.4 shows the S2 sustainability indicator according to the COM no-policy-change scenario. For a significant number of countries (11 Member States) both the CoA and the IBP components contribute to the fiscal gap. Finally,

^{(&}lt;sup>38</sup>) As described in section 4.2, the SCP scenario is based on the macroeconomic and fiscal projections contained in Member States' fiscal plans. Beyond the SCP horizon, fiscal policy (before the adjustment as implied by the S1 indicator) is assumed to remain unchanged over the rest of the projection horizon.



the diagonal lines indicate the size of the sustainability gap. Among the Member States, 2 countries (LU and RO) face *high* risk according to the S2 indicator, 14 countries (BE, CZ, IE, ES, IT, HU, MT, NL, AT,

Note: The graph shows the initial budgetary position (IBP) on the horizontal axis and the long-term change in the fiscal position due to cost of ageing (CoA) on the vertical axis. A position to the left indicates a favourable IBP (i.e. if it is below zero, it means that the budgetary position supports fiscal sustainability). A position towards the bottom of the axis indicates a low long-term 'cost of ageing'. **Source:** European Commission

PL, SI, SK, FI and UK) face *medium* risk, and 11 countries face *low* risk over the long-term. When considering in addition the results of the DSA overall risk classification, long-term risks are found to be one step higher in 8 countries, either from *low* to *medium* (FR, HR, CY and PT), or from *medium* to *high* (BE, ES, IT, and UK) (see Table 4.1). (³⁹)

^{(&}lt;sup>39</sup>) For details about the revised methodology behind the long-term sustainability risk classification, see Box 4.1 in European Commission (2019), 'Fiscal Sustainability Report 2018', *European Economy, Institutional papers*, No 94, EC, Brussels.



Graph 4.5: The S2 sustainability gap: 'COM no-policy-change' and 'SCP' scenarios (in pps. of GDP)

Developments assuming implementation of the fiscal plans in the SCPs

Even assuming the full implementation of the fiscal plans in the SCPs, 13 Member States would still face significant sustainability gaps. Graph 4.5 shows the S2 indicator with two different starting points: (i) the COM no-policy-change scenario and (ii) the SCP scenario. The SCP scenario shows the extent to which the implementation of the governments' fiscal plans would contribute to ensuring fiscal sustainability. Under the assumption that the fiscal plans in the programmes are fully implemented, a large majority of Member States are expected to have a lower sustainability gap (as shown by a position below the 45° degrees line in the graph). In the EU as a whole, the S2 fiscal gap would reach 1.4 pps. of GDP, i.e. slightly above half of the gap in the COM no-policy-change scenario. Even assuming the full implementation of the fiscal plans in the SCPs, 13 Member States would still have sustainability gaps of at least 2 pps. of GDP. In terms of risk classification, in the SCP scenario, 6 Member States would go to a lower risk category (IE, ES, IT, HU and PL from *medium* to *low* risk and RO from *high* to *medium* risk). In the cases of BG, CZ, DK, DE, EE, CY, NL, and AT, the S2 indicator, based on the SCPs, would reach a higher value than based on the Commission no-policy-change scenario, notably as the result of the assumed lower structural primary surplus. However, this would lead to an unfavourable change in the risk category only in the cases of BG and DE (from *low* to *medium* risk).

Table 4.1:	Overall risk classification in the 2019 assessment round, COM no-policy-change scenario												
	S0 Overall SHORT-TERM risk category	Overall MEDIUM-TERM risk category	S1 indicator - overall risk assessment	Debt sustainability analysis - overall risk assessment	S2 indicator - overall risk assessment	Overall LONG-TERM risk category							
BE	LOW (0.22)	HIGH	HIGH (4.7)	HIGH	MEDIUM (4.3)	HIGH							
BG	LOW (0.27)	LOW	LOW (-4.6)	LOW	LOW (1.5)	LOW							
cz	LOW (0.19)	LOW	LOW (-2.1)	LOW	MEDIUM (4.7)	MEDIUM							
DK	LOW (0.12)	LOW	LOW (-5.5)	LOW	LOW (-0.9)	LOW							
DE	LOW (0)	LOW	LOW (-1.5)	LOW	LOW (1.9)	LOW							
EE	LOW (0.2)	LOW	LOW (-3.4)	LOW	LOW (1.6)	LOW							
IE	LOW (0.21)	LOW	LOW (-0.5)	LOW	MEDIUM (3.5)	MEDIUM							
ES	LOW (0.37)	HIGH	HIGH (5.6)	HIGH	MEDIUM (2.4)	HIGH							
FR	LOW (0.24)	HIGH	HIGH (5)	HIGH	LOW (0.4)	MEDIUM							
HR	LOW (0.24)	MEDIUM	LOW (-0.5)	MEDIUM	LOW (-2.4)	MEDIUM							
IT	LOW (0.36)	HIGH	HIGH (10.2)	HIGH	MEDIUM (3.2)	HIGH							
CY	HIGH (0.54)	MEDIUM	LOW (-0.7)	MEDIUM	LOW (-0.7)	MEDIUM							
LV	LOW (0.24)	LOW	LOW (-2.2)	LOW	LOW (0.6)	LOW							
LT	LOW (0.21)	LOW	LOW (-1.2)	LOW	LOW (1)	LOW							
LU	LOW (0.12)	LOW	LOW (-4.6)	LOW	HIGH (8.4)	HIGH							
HU	LOW (0.28)	MEDIUM	MEDIUM (0.5)	MEDIUM	MEDIUM (3.9)	MEDIUM							
МТ	LOW (0.06)	LOW	LOW (-4.8)	LOW	MEDIUM (3.4)	MEDIUM							
NL	LOW (0.08)	LOW	LOW (-2)	LOW	MEDIUM (2.7)	MEDIUM							
AT	LOW (0.03)	LOW	LOW (-1.1)	LOW	MEDIUM (2.4)	MEDIUM							
PL	LOW (0.18)	MEDIUM	MEDIUM (0.7)	MEDIUM	MEDIUM (3.5)	MEDIUM							
PT	LOW (0.33)	HIGH	HIGH (3.7)	HIGH	LOW (0.3)	MEDIUM							
RO	LOW (0.3)	HIGH	MEDIUM (2.2)	HIGH	HIGH (6.3)	HIGH							
SI	LOW (0.13)	LOW	LOW (-0.7)	LOW	MEDIUM (4.8)	MEDIUM							
SK	LOW (0.27)	LOW	LOW (-1.9)	LOW	MEDIUM (3.2)	MEDIUM							
FI	LOW (0.14)	LOW	LOW (0)	LOW	MEDIUM (2.5)	MEDIUM							
SE	LOW (0.2)	LOW	LOW (-4.4)	LOW	LOW (1.2)	LOW							
UK	LOW (0.37)	HIGH	MEDIUM (1.8)	HIGH	MEDIUM (3.3)	HIGH							
EU	:	:	MEDIUM (1.9)	:	LOW (2.3)	:							
EA	:	:	HIGH (2.6)	:	MEDIUM (2)	:							

S0 indicator: The critical threshold for the overall S0 indicator is 0.46.

The S1 indicator: The following thresholds for the S1 indicator are used:

- if the S1 value is less than zero, the country is assigned low risk.

- if a structural adjustment in the primary balance of up to 0.5 p.p. of GDP per year for five years after the last year covered is required (indicating an cumulated adjustment of 2.5 pp.), it is assigned medium risk. - if it is greater than 2.5 (meaning a structural adjustment of more than 0.5 p.p. of GDP per year is necessary), it is assigned

high risk.

The Debt Sustainability Analysis (DSA): a range of factors (such as debt levels, debt paths, alternative underlying assumptions, stochastic projections) are used for the risk classification. For details about the sustainability risk classification and the methodology behind the Debt Sustainability Analysis (DSA), see European Commission (2019), 'Fiscal Sustainability Report

2018', European Economy, Institutional papers, No 94, EC, Brussels.

The S2 indicator: The following thresholds for the S2 indicator are used:

- if the value of \$2 is lower than 2, the country is assigned low risk.

- if it is between 2 and 6, it is assigned medium risk.

- if it is greater than 6, it is assigned high risk.

Source: European Commission

Box 4.2: Fiscal sustainability under market-based interest rate expectations assumptions

The rationale for considering an alternative interest rate assumption is to account better for the currently low and diverse interest rate environment. Financial assumptions are a key input of DSA frameworks, along with fiscal and macroeconomic assumptions. The baseline market interest rate assumptions entail a common convergence of long-term (short-term) interest rates to 5% (4%) by 2029. This implies that under the baseline, current interest rate conditions are reflected only at the beginning of the projection period by relying on actual data as a starting point for the projection, while a gradual convergence to the 5% common target is driving the (common) interest rate path over the medium-run.

Relying on market-based interest rate expectations to set country-specific convergence targets allows reflecting the current environment beyond the short-term (¹). Such market-based interest rate expectations' estimates are based on standard forward rate computations. Concretely, alternative targets for long-term interest rates are set equal to forward long-term interest rates ($lti_{10,t+10}$) at t+10, computed as follows (²):

$$\left(1 + lti_{10,t+10}\right)^{10} = \frac{\left(1 + lti_{20,t}\right)^{20}}{\left(1 + lti_{10,t}\right)^{10}}$$

where $lti_{20,t}$ and $lti_{10,t}$ stand for the 10 and 20 years maturity spot yields and $lti_{10,t+10}$ stands for the (forward) 10 years maturity yield at t+10. Computations use latest available market (monthly averaged) data – i.e. April 2019.

In turn, short-term interest rate targets are computed by applying a 0.8 scaling factor to the long-term interest rate targets. This factor reflects the standard slope of the euro area yield curve, a factor also retained under the current baseline – i.e. short-term baseline target (4%) is equal to 0.8 times the long-term baseline target (5%) $\binom{3}{2}$.

The resulting market-based interest rate targets imply a less sharp increase in interest rates and less convergence across countries over the projection horizon. Table 1 reveals that long-term interest rate targets are lower than under the baseline (5%) for all countries, except for Romania. The impact on short-term rates targets is similar, as those are set as a fraction of the long-term interest rate targets. However, in some cases however, notably Italy, the market-based long-term rate target remains not so far from the 5% assumed under the baseline.

The impact of using market-based interest rate targets on sustainability analysis varies across countries but leaves the risk assessment unchanged in all cases. Large impacts on the projected debt levels or on the S1 indicator concern countries with relatively high debt (e.g. IT, ES, PT) or with financing conditions expected, based on market data, to be substantially lower than those assumed under the baseline (e.g. DE, NL), while in some countries both aspects matter (e.g. BE, FR). Instead, in all cases, the S2 indicator appears little affected. This reflects the fact that this indicator relies on a substantially longer horizon, beyond T+10, which dampens

^{(&}lt;sup>1</sup>) ¹) For a detailed discussion see Box 3.2 entitled "Using financial markets' interest rate expectations to project interest rates: does it change the risk assessment?", in *The 2018 Fiscal Sustainability Report*, European Economy Institutional Paper 094, January 2019.

⁽²⁾ Moreover, for countries for which yield curve data are missing, we rely on the following assumptions:

⁻CY, LT: the long-term spread vis-à-vis DE follows the same pattern as the average spread across available euro area countries (excl. DE & EL);

⁻BG, HR: the long term spread vis-à-vis DE follows the same pattern as the average spread across available non-euro area countries;

⁻EE: the forward interest rate is the average of the forward interest rate of LV and LT.

^{(&}lt;sup>3</sup>) In this scenario, long-term (short-term) interest rates are further assumed to gradually converge to 5% (4%) by T+30, beyond 2029.

Box (continued)

which dampens the impact of change in assumption affecting data only up to T+10 (⁴)). Table 1 also shows overall no impact on the risk assessment. Risk categories remain unchanged, except in the case of Cyprus. For the latter, a slight reduction in the projected debt-to-GDP by 2029, under the baseline, leads to a change in category for this criterion, reflecting a threshold effect. Yet, the overall assessment for Cyprus would remain unchanged as other scenarios and criteria continue to point at medium risk, even under the market-based interest rate assumption.

Graph 1 shows that for high debt countries (BE, CY, ES, FR, IT and PT) the impact of relying on this alternative interest rate assumption is not negligible, yet confirms the slight upward trend projected under a no-fiscal policy change assumption. The profile of the debt-to-GDP ratio projection for those countries posts a more benign surge by the end of the projection horizon under the market-based interest rate assumption. At the same time, even under the more favourable interest rate assumption used in this scenario, important debt-vulnerabilities would remain over the medium-term for those countries, calling for additional policy measures. A close monitoring of financial markets' developments is important. In particular, a further easing, or reversal of recent deterioration, in financing costs faced by some high debt countries would help ease even further the debt profile reported for those countries under the market-based interest rate assumption.

Overall, in terms of policy implications, the results point at an overall moderate impact of using interest rates' assumptions based on financial market expectations – notably in terms of risk classification – although impacts are larger for some countries. The latter are those (e.g. FR, BE) that tend to have higher debt, shorter average maturity and for which market expectations are particularly benign, substantially deviating from the assumed convergence, by the end of the horizon, to a common 5% nominal long-term interest rate target level (⁵) (e.g. DE, NL). Overall, such differentiated sensitivity vis-à-vis interest rate changes confirms the need for additional policy measures in high-debt countries, which could in turn help easing financial markets' pressures in some cases, and reduce further fiscal sustainability risks.

(Continued on the next page)

^{(&}lt;sup>4</sup>) The impact on the S2 indicator is also ambiguous as opposing effect are at play. While a lower interest rate assumption reduces the burden of the debt, it also reduces the discounting of the future cost of ageing, an important component in the S2 computations. Additionally, the level at which debt-to-GDP stabilises differs under the two interest rate assumptions. Overall, this suggest the need for caution when interpreting the small changes in the S2 indicator reported here.

^{(&}lt;sup>5</sup>) For further discussions on differentiated sensitivities vis-à-vis interest rate changes, see also Box 2.2 entitled "The sensitivity of public debt to a rise in interest rates in EU countries", in *The 2017 Debt Sustainability Monitor*, European Economy Institutional Paper 071, January 2018.

Box (continued)

Table 1: Impact of alternative interest rate assumptions on debt-to-GDP, \$1, \$2 and long-term interest	t rate targets, in EU countries
--	---------------------------------

	Debt-to-GDP				S1 indicator			S2 indicator		Market-based	1
	Baseline	Alternative	Impact	Baseline	Alternative	Impact	Baseline	Alternative	Impact		
BE	103	96	-7.1	4.7	3.8	-0.9	4.3	4.2	-0.2	1.9	BE
BG	8	8	-0.2	-4.6	-5.4	-0.8	1.5	1.5	0.1	1.5	BG
cz	32	30	-1.4	-2.1	-2.5	-0.4	4.7	4.7	0.0	3.4	cz
DK	9	8	-1.1	-5.5	-6.5	-1.0	-0.9	-0.9	0.0	0.9	DK
DE	42	38	-4.6	-1.5	-2.6	-1.1	1.9	1.8	-0.1	1.1	DE
EE	16	15	-1.0	-3.4	-4.1	-0.7	1.6	1.6	0.0	2.1	EE
IE	50	47	-3.2	-0.5	-1.2	-0.7	3.5	3.5	0.0	2.0	IE
ES	111	105	-5.6	5.6	5.0	-0.7	2.4	2.2	-0.2	2.9	ES
FR	106	98	-7.9	5.0	4.0	-1.0	0.4	-0.1	-0.5	1.7	FR
HR	59	55	-3.6	-0.5	-1.2	-0.7	-2.4	-2.7	-0.3	2.8	HR
IT	153	149	-4.8	10.2	9.7	-0.5	3.2	2.9	-0.3	4.0	IT
CY	61	59	-2.4	-0.7	-1.3	-0.6	-0.7	-0.9	-0.2	3.0	CY
LV	32	31	-1.8	-2.2	-2.8	-0.6	0.6	0.5	-0.1	2.4	LV
LT	38	35	-3.1	-1.2	-2.0	-0.8	1.0	0.9	-0.2	1.8	LT
LU	10	10	-0.1	-4.6	-5.1	-0.6	8.4	8.7	0.3	1.8	LU
HU	64	61	-2.7	0.5	0.1	-0.5	3.9	3.8	0.0	3.7	HU
мт	17	16	-0.7	-4.8	-5.4	-0.6	3.4	3.5	0.1	2.4	мт
NL	36	33	-3.3	-2.0	-3.0	-1.0	2.7	2.7	0.0	0.9	NL
AT	49	46	-2.8	-1.1	-1.8	-0.7	2.4	2.4	0.0	1.6	AT
PL	58	56	-2.1	0.7	0.4	-0.4	3.5	3.4	0.0	3.6	PL
PT	102	98	-4.2	3.7	3.2	-0.6	0.3	0.0	-0.3	3.1	PT
RO	66	67	0.7	2.2	2.3	0.1	6.3	6.3	0.0	5.5	RO
SI	47	44	-2.7	-0.7	-1.4	-0.6	4.8	4.9	0.0	2.4	SI
SK	40	38	-1.8	-1.9	-2.5	-0.6	3.2	3.2	0.0	2.1	sк
FI	56	54	-2.3	0.0	-0.6	-0.5	2.5	2.5	0.0	1.3	FI
SE	17	17	-0.4	-4.4	-5.0	-0.6	1.2	1.2	0.0	1.6	SE
UK	78	74	-4.5	1.8	1.1	-0.7	3.3	3.2	-0.1	2.2	UK

(1) Colours in the table correspond to fiscal sustainability risk categories: low (green), medium (yellow) and high (red). Source: European Commission



Graph 1. Medium-term government debt projections for high-debt countries (in % of GDP)

ANNEX 1

		2019 Stabil	ity and Con	vergence P	rogrammes		Commi	eeion 2019	epring for	ecast	Difference	compa means	red to
		2019 01401	ity and con	vergence i	rogrammea		Commi	551011 2013	spring ion	ecasi	programme)	means	inglier in
	201	8 2010	2020	2021	2022	2023	2017	2018	2010	2020	2018	2010	2020
B	E 14	1 13	3 14	14	12	n a	17	1.4	12	12	0.0	0.1	0.2
	Y 39	3 37	7 32	3.0	3.0	n a	4.5	3.9	31	27	0.0	0.1	0.5
	E 12/4	1 1	1 2/4	1	1	1	2.2	1.4	0.5	1.5	0.0	0.5	0.1
E	E 3.9	9 3.1	2.7	2.7	2.6	2.5	4.9	3.9	2.8	2.4	0.0	0.3	0.3
E	L 1.9	2.3	3 2.3	2.1	2.0	n.a.	1.5	1.9	2.2	2.2	0.0	0.1	0.1
	IE 6.	7 3.9	3.3	2.4	2.5	2.6	7.2	6.7	3.8	3.4	0.0	0.1	-0.1
E	S 2.0	6 2.2	2 1.9	1.8	1.8	n.a.	3.0	2.6	2.1	1.9	0.0	0.1	0.1
F	R 1.6	6 1.4	1 1.5	1.4	1.4	n.a.	2.2	1.6	1.3	1.5	0.0	0.1	0.0
'	IT 0.9	9 0.2	2 0.8	0.8	0.8	n.a.	1.7	0.9	0.1	0.7	0.0	0.1	0.1
L	.V 4.8	3 3.2	2 3.0	2.9	2.9	n.a.	4.6	4.8	3.1	2.8	0.0	0.1	0.2
L	.T 3.4	4 2.6	6 2.4	2.3	2.3	n.a.	4.1	3.4	2.7	2.4	0.0	-0.1	0.0
L	.U 3.0) 3.0) 3.8	3.5	3.0	2.5	1.5	2.6	2.5	2.6	0.4	0.5	1.2
M	IT 6.6	6.2	2 5.7	5.1	5.0	n.a.	6.7	6.6	5.5	4.8	0.0	0.7	0.9
N	IL 2.7	7 1.5	5 1.5	1.2	1.0	n.a.	2.9	2.7	1.6	1.6	0.0	-0.1	-0.1
A	T 2.1	7 1.7	7 1.8	1.7	1.6	1.4	2.6	2.7	1.5	1.6	0.0	0.2	0.1
P	PT 2.*	1 1.9	9 1.9	2.0	2.0	2.1	2.8	2.1	1.7	1.7	0.0	0.2	0.2
5	SI 4.	5 3.4	4 3.1	2.8	2.7	n.a.	4.9	4.5	3.1	2.8	0.0	0.3	0.3
S	SK 4.1	1 4.0) 3.7	3.2	2.5	n.a.	3.2	4.1	3.8	3.4	0.0	0.3	0.3
	FI 2.3	3 1.7	7 1.4	1.2	1.0	n.a.	2.7	2.3	1.6	1.2	0.0	0.1	0.2
E/	A 1.9	9 1.4	1 1.6	1.4	1.3	n.a.	2.4	1.9	1.2	1.5	0.0	0.2	0.1
В	G 3.	1 3.4	1 3.3	3.3	3.3	n.a.	3.8	3.1	3.3	3.4	0.0	0.1	-0.1
	2	3 2.4	1 2.4	2.3	2.2	n.a.	4.4	2.9	2.6	2.4	0.0	-0.2	0.0
U U	K 1.	1.1	1.6	1.5	1.1	n.a.	2.3	1.4	1.7	1.6	0.3	0.0	0.0
	R 2.0	5 Z.C	2.4	2.3	2.3	n.a.	2.9	2.0	2.0	2.5	0.0	-0.1	-0.1
п	4.3	9 4.0		4.1	4.2	4.0	4.1	4.9	3.7	2.8	0.0	0.3	1.2
R F	4.	1 5.0	5 - 5.7	5.0	5.0	n.a.	7.0	4.1	3.3	3.1	0.0	2.2	2.0
	L 3.	1 4.0 2 1.6	3.7	3.4	3.3	n.a.	4.0	2.1	4.2	3.0	0.0	-0.2	0.1
		5 I.C	5 1.0 5 1.4	1.0	2.0	1.a. 1.6	2.1	2.3	1.4	1.0	0.0	0.2	0.0
F	1. 20	$\frac{1.2}{1.2}$	5 17	1.0	1.0	n.o	2.5	2.0	1.1	1.6	-0.1	0.1	0.3

Table A1.1: Real GDP growth (%)

Note: EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

Table A1.2	: Output	t gap (%	of pot. Gl	DP)									
	201	19 Stability	and Conve	ergence Pro	ogrammes		Commis	sion 2019 s	spring fored	ast	Difference forecast (re programme)	compare d means h	ed to ligher in
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	0.2	0.2	0.2	0.2	0.2	0.0	0.3	0.2	0.2	0.1	0.0	0.0	0.1
CY	2.1	3.0	3.1	3.1	3.2	0.0	0.8	2.8	3.8	4.3	-0.6	-0.7	-1.2
DE	0.5	0.1	0.3	0.1	0.0	0.0	0.9	0.7	-0.2	-0.1	-0.2	0.3	0.4
EE	3.2	2.9	2.6	2.6	2.7	2.8	2.7	3.3	2.8	2.2	-0.1	0.1	0.5
EL	-6.4	-4.1	-2.1	-0.5	0.8	0.0	-8.9	-6.4	-4.0	-1.9	0.0	-0.1	-0.2
IE	2.8	2.4	1.7	0.5	-0.2	-0.8	1.2	2.8	2.2	1.7	0.1	0.2	0.0
ES	0.8	1.4	1.6	1.9	2.2	0.0	-0.5	0.9	1.6	2.0	-0.1	-0.2	-0.5
FR	0.3	0.4	0.5	0.6	0.8	0.0	0.0	0.4	0.5	0.7	0.0	-0.1	-0.2
IT	-0.2	-0.3	-0.2	-0.1	0.0	0.0	-0.5	-0.1	-0.3	-0.1	0.0	0.0	-0.1
LV	2.9	2.1	0.9	0.2	-0.2	0.0	1.7	2.8	2.2	1.3	0.0	-0.1	-0.3
LT	3.7	2.7	1.4	0.5	0.0	0.0	3.0	3.6	3.1	2.1	0.0	-0.3	-0.7
LU	0.7	1.0	1.6	1.8	1.8	1.5	-0.2	0.6	1.0	1.3	0.1	-0.1	0.3
MT	0.8	0.7	0.1	-0.1	0.0	0.0	1.1	1.3	1.1	0.3	-0.5	-0.4	-0.3
NL	1.1	0.9	0.7	0.4	0.1	0.0	0.2	1.1	1.0	0.9	0.0	-0.1	-0.2
AT	0.9	0.7	0.6	0.2	0.1	-0.1	0.1	1.0	0.7	0.4	-0.1	0.0	0.1
PT	0.9	0.9	0.6	0.5	0.7	0.8	0.7	1.2	1.3	1.4	-0.3	-0.4	-0.8
SI	3.3	3.4	3.0	1.9	0.6	0.0	1.3	3.3	3.4	2.8	0.1	0.0	0.2
SK	1.2	1.8	1.9	1.6	0.9	0.0	0.4	1.5	2.1	2.3	-0.3	-0.3	-0.4
FI	0.5	0.6	0.4	0.5	0.5	0.0	-0.3	0.7	0.9	0.8	-0.1	-0.3	-0.4
EA	0.4	0.4	0.5	0.4	0.5	0.0	0.1	0.5	0.4	0.5	-0.1	0.0	0.0
BG	0.2	0.3	0.4	0.5	0.8	0.0	0.1	0.2	0.6	1.3	0.1	-0.3	-0.9
CZ	1.3	0.7	0.3	0.1	0.0	0.0	1.3	1.2	0.8	0.3	0.0	-0.1	0.0
DK	-0.5	-0.4	-0.3	-0.1	-0.3	0.0	-0.1	-0.5	-0.5	-0.7	-0.1	0.2	0.4
HR	1.4	1.8	1.8	1.7	1.9	0.0	0.5	1.2	2.1	2.4	0.2	-0.3	-0.6
HU	3.0	2.5	2.1	1.6	1.3	1.0	1.8	3.3	3.3	2.5	-0.4	-0.8	-0.4
RO	-0.4	0.1	0.7	0.7	0.9	0.0	0.8	0.9	0.6	0.2	-1.4	-0.5	0.4
PL	2.2	2.2	1.8	1.3	0.8	0.0	0.8	2.1	2.3	2.0	0.1	-0.1	-0.2
SE	0.4	0.0	-0.4	-0.6	-0.3	0.0	0.3	0.5	-0.1	-0.4	-0.1	0.1	0.0
UK	0.9	0.3	0.0	0.0	0.0	-0.1	1.0	0.7	0.5	0.3	0.1	-0.1	-0.2
E11	0.6	0.4	0.4	0.4	0.4	0.0	0.2	0.6	0.4	0.5	0.1	0.0	0.1

Note: For SCPs: recalculated by the European Commission on the basis of the information in the programme according to the commonly-agreed methodology. EU averages are based on UK Fiscal year data. **Source:** European Commission based on Commission 2019 spring forecast and 2019 SCPs

10	DIE AT.3:	Genero	ii govern	ment bai	ance (%	of GDP)								
		201	9 Stability	and Conve	rgence Pro	grammes		Commiss	sion 2019 s	ast	Difference compared to forecast (red means higher in programme)			
		2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
	BE	-0.7	-0.8	-0.2	0.1	0.0	n.a.	-0.8	-0.7	-1.3	-1.5	0.0	0.5	1.3
	CY	-4.8	3.0	2.6	2.4	2.2	n.a.	1.8	-4.8	3.0	2.8	0.0	0.0	-0.3
	DE	1 3/4	3/4	3/4	2/4	2/4	0.6	1.0	1.7	1.0	0.8	0.0	-0.1	0.0
	EE	-0.5	-0.2	-0.3	-0.5	-0.7	-0.2	-0.4	-0.6	-0.3	-0.5	0.1	0.1	0.2
	EL	1.1	1.6	1.1	1.3	1.7	n.a.	0.7	1.1	0.5	-0.1	0.0	1.1	1.2
	IE	0.0	0.2	0.4	0.7	1.0	1.3	-0.3	0.0	0.0	0.3	0.0	0.2	0.1
	ES	-2.5	-2.0	-1.1	-0.4	0.0	n.a.	-3.1	-2.5	-2.3	-2.0	0.0	0.3	0.9
	FR	-2.5	-3.1	-2.0	-1.6	-1.2	n.a.	-2.8	-2.5	-3.1	-2.2	0.0	0.0	0.2
	IT	-2.1	-2.4	-2.1	-1.8	-1.5	n.a.	-2.4	-2.1	-2.5	-3.5	0.0	0.1	1.4
	LV	-1.0	-0.5	-0.4	-0.2	-0.3	n.a.	-0.6	-1.0	-0.6	-0.6	0.0	0.1	0.2
	LT	0.7	0.4	0.2	0.1	0.1	n.a.	0.5	0.7	0.3	0.0	0.0	0.1	0.2
	LU	2.6	1.0	1.4	1.5	2.0	2.2	1.4	2.4	1.4	1.1	0.2	-0.4	0.2
	MT	2.0	0.9	1.0	1.1	1.1	n.a.	3.4	2.0	1.1	0.9	0.0	-0.2	0.1
	NL	1.5	1.2	0.8	0.2	0.0	n.a.	1.2	1.5	1.4	0.8	0.0	-0.2	0.0
	AT	0.1	0.3	0.2	0.2	0.0	0.0	-0.8	0.1	0.3	0.2	0.0	0.0	0.0
	PT	-0.5	-0.2	0.3	0.9	0.7	0.7	-3.0	-0.5	-0.4	-0.1	0.0	0.2	0.4
	SI	0.7	0.9	1.0	1.1	1.2	n.a.	0.0	0.7	0.7	0.9	0.0	0.3	0.0
	SK	-0.7	0.0	0.0	0.0	0.0	n.a.	-0.8	-0.7	-0.5	-0.6	0.0	0.5	0.6
	FI	-0.6	-0.3	0.0	-0.1	-0.3	n.a.	-0.8	-0.7	-0.4	-0.2	0.1	0.1	0.1
	EA	-0.5	-0.9	-0.5	-0.4	-0.3	n.a.	-1.0	-0.5	-0.9	-0.9	0.0	0.0	0.4
	BG	2.0	-0.3	0.4	0.2	0.1	n.a.	1.2	2.0	0.8	1.0	0.0	-1.1	-0.6
	CZ	0.9	0.3	-0.2	-0.3	-0.5	n.a.	1.6	0.9	0.2	-0.2	0.0	0.1	0.0
	DK	0.2	-0.1	-0.1	-0.2	0.0	n.a.	1.4	0.5	0.6	-0.1	-0.3	-0.6	-0.1
	HR	0.2	-0.3	0.2	0.4	0.8	n.a.	0.8	0.2	0.1	0.5	0.0	-0.5	-0.4
	HU	-2.2	-1.8	-1.5	-1.2	-0.5	0.0	-2.2	-2.2	-1.8	-1.6	0.0	0.0	0.1
	RO	-3.0	-2.8	-2.7	-2.4	-2.0	n.a.	-2.7	-3.0	-3.5	-4.7	0.0	0.7	2.0
	PL	-0.4	-1.7	0.2	-0.3	-0.6	n.a.	-1.5	-0.4	-1.6	-1.4	0.0	-0.1	1.6
	SE	0.7	0.6	0.7	1.1	1.9	n.a.	1.4	0.9	0.4	0.4	-0.2	0.2	0.3
	UK	-1.2	-1.4	-1.1	-1.1	-0.7	-0.6	-2.1	-1.3	-1.5	-1.2	0.1	0.1	0.1
	EU	-0.6	-0.9	-0.6	-0.5	-0.3	n.a.	-1.0	-0.6	-1.0	-1.0	0.0	0.0	0.4

Table A1.3: Ge al c at bala (% of GDP)

Note: EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

	20 [.]	19 Stability	and Conve	ergence Pro	ogrammes		Commis	sion 2019 s	spring fored	ast	Difference compared forecast (red means higher programme)		
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	51.7	51.1	51.3	51.3	51.3	n.a.	51.3	52.4	52.3	52.5	-0.7	-1.2	-1.2
CY	39.7	40.7	40.9	40.6	40.1	n.a.	39.1	44.7	38.0	39.0	-5.0	2.7	1.9
DE	45 2/4	45 2/4	45 1/4	45	45	45 1/4	45.0	43.9	44.6	44.7	1.7	0.8	0.6
EE	39.1	40.0	39.5	38.8	37.8	37.1	38.9	39.5	39.9	39.9	-0.4	0.1	-0.4
EL	48.0	47.8	45.9	45.1	44.2	n.a.	48.1	46.7	46.7	45.6	1.3	1.1	0.3
IE	25.8	25.6	25.2	25.1	25.1	n.a.	26.0	25.7	25.6	25.0	0.1	0.0	0.2
ES	38.9	39.1	39.8	40.3	40.7	n.a.	37.9	41.3	41.2	41.0	-2.5	-2.1	-1.1
FR	53.5	52.4	52.3	52.0	51.7	n.a.	53.6	56.0	55.5	54.4	-2.5	-3.1	-2.1
IT	46.4	46.5	47.1	47.0	46.6	n.a.	46.5	48.6	49.2	49.8	-2.2	-2.7	-2.7
LV	37.5	36.2	35.9	35.4	34.9	n.a.	37.2	38.5	37.8	37.4	-1.0	-1.6	-1.5
LT	34.7	36.0	36.0	35.9	35.6	n.a.	33.6	34.0	34.4	34.9	0.7	1.6	1.1
LU	45.9	45.0	45.0	45.0	45.0	45.2	44.4	43.1	43.8	44.3	2.8	1.1	0.7
MT	38.8	38.1	36.7	35.7	35.2	n.a.	39.2	36.8	37.1	37.1	2.0	0.9	-0.4
NL	43.8	44.3	43.9	43.3	43.2	n.a.	43.7	42.2	42.6	42.8	1.6	1.7	1.1
AT	48.6	48.3	47.7	47.1	46.6	46.3	48.4	48.5	48.0	47.7	0.1	0.2	0.0
PT	43.5	43.8	43.7	43.9	43.2	43.0	42.7	44.0	44.2	43.9	-0.5	-0.5	-0.3
SI	43.1	43.2	42.6	41.8	41.4	n.a.	43.2	42.4	42.5	42.1	0.7	0.7	0.5
SK	39.9	39.1	38.4	37.9	37.7	n.a.	39.4	40.6	40.4	40.2	-0.7	-1.3	-1.8
FI	52.5	52.3	52.2	52.0	51.9	n.a.	53.4	53.1	52.8	52.6	-0.6	-0.5	-0.4
EA	46.3	46.0	46.0	45.8	45.6	n.a.	46.1	46.8	47.0	46.8	0.5	-0.9	-0.7
BG	36.8	37.7	37.3	36.9	36.1	n.a.	36.2	34.8	37.0	36.7	2.0	0.7	0.6
CZ	41.7	41.5	41.3	41.2	41.2	n.a.	40.5	40.8	41.7	41.9	0.9	-0.2	-0.6
DK	50.5	49.6	49.1	49.1	49.2	n.a.	52.6	51.4	51.1	51.5	-0.9	-1.5	-2.4
HR	46.6	47.0	46.9	46.4	46.2	n.a.	46.1	46.4	46.7	46.1	0.2	0.3	0.7
HU	44.2	44.6	43.3	41.9	40.2	39.4	44.7	46.5	46.4	45.6	-2.3	-1.8	-2.3
RO	32.0	33.8	33.8	33.5	33.9	n.a.	30.9	35.0	36.1	38.0	-3.0	-2.3	-4.2
PL	41.2	41.7	43.0	41.8	40.7	n.a.	39.7	41.5	42.6	42.6	-0.3	-0.9	0.4
SE	50.5	49.9	49.7	49.6	49.4	n.a.	50.9	49.9	49.8	49.5	0.6	0.1	0.2
UK	37.0	36.8	36.9	36.7	37.0	37.1	38.6	38.9	39.1	39.4	-1.9	-2.3	-2.5
EU	44.6	44.4	44.4	44.2	44.0	n.a.	44.7	45.3	45.5	45.4	-0.7	-1.1	-1.0

Table A1.4: General government total revenue (% of GDP)

Note: EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

Table A1.5	: Gener	al govern	iment tot	al expen	diture (%	of GDP)						
	20	19 Stability	and Conv	ergence Pr	ogrammes		Commission 2019 spring forecast				Difference compared to forecast (red means higher in programme)		
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	52.4	51.9	51.4	51.3	51.3	n.a.	52.1	52.4	52.3	52.5	0.0	-0.4	-1.1
CY	44.5	37.7	38.3	38.2	38.0	n.a.	37.4	44.7	38.0	39.0	-0.3	-0.3	-0.7
DE	43 3/4	44 2/4	44 2/4	44 2/4	44 2/4	44 2/4	43.9	43.9	44.6	44.7	0.0	0.0	-0.1
EE/	39.5	40.2	39.7	39.3	38.5	37.3	39.3	39.5	39.9	39.9	0.0	0.3	-0.2
EL/	46.9	46.2	44.8	43.8	42.5	n.a.	47.3	46.7	46.7	45.6	0.2	-0.5	-0.8
IE!	25.7	25.4	24.8	24.4	24.1	n.a.	26.3	25.7	25.6	25.0	0.0	-0.2	-0.2
ES	41.3	41.1	41.0	40.7	40.7	n.a.	41.0	41.3	41.2	41.0	0.0	-0.1	0.0
FR	56.0	55.5	54.3	53.6	53.0	n.a.	56.4	56.0	55.5	54.4	0.0	0.0	-0.1
IT	48.6	48.9	49.1	48.8	48.5	n.a.	48.9	48.6	49.2	49.8	0.0	-0.3	-0.7
LV	38.5	36.7	36.4	35.6	35.2	n.a.	37.8	38.5	37.8	37.4	0.0	-1.1	-1.0
LT	34.0	35.5	35.9	35.9	35.5	n.a.	33.1	34.0	34.4	34.9	0.0	1.1	1.0
LU	43.3	43.9	43.6	43.5	43.0	43.0	43.0	43.1	43.8	44.3	0.2	0.1	-0.7
MT	36.8	37.2	35.7	34.7	34.1	n.a.	35.7	36.8	37.1	37.1	0.0	0.0	-1.4
NL	42.2	43.0	43.1	43.1	43.3	n.a.	42.5	42.2	42.6	42.8	0.0	0.4	0.3
AT	48.5	47.9	47.5	46.9	46.6	46.3	49.2	48.5	48.0	47.7	0.0	-0.1	-0.2
PT	44.0	43.9	43.4	43.0	42.6	42.4	45.7	44.0	44.2	43.9	0.0	-0.3	-0.5
SI	42.4	42.2	41.6	40.7	40.2	n.a.	43.2	42.4	42.5	42.1	0.0	-0.2	-0.4
SK	40.6	39.1	38.4	37.9	37.7	n.a.	40.2	40.6	40.4	40.2	. 0.0	-1.3	-1.8
FI	53.1	52.6	52.2	52.0	52.2	n.a.	54.2	53.1	52.8	52.6	0.0	-0.3	-0.4
EA	46.8	46.9	46.5	46.2	46.0	n.a.	47.0	46.8	47.0	46.8	0.0	-0.1	-0.3
BG	34.8	38.0	36.9	36.7	36.0	n.a.	35.0	34.8	37.0	36.7	0.0	1.0	0.2
CZ	40.8	41.2	41.5	41.6	41.7	n.a.	38.9	40.8	41.7	41.9	0.0	-0.5	-0.4
DK	50.3	49.7	49.2	49.3	49.2	n.a.	51.2	51.4	51.1	51.5	-1.1	-1.4	-2.3
HR	46.4	47.3	46.7	46.0	45.4	n.a.	45.3	46.4	46.7	46.1	0.0	0.6	0.6
HU	46.5	46.4	44.8	43.1	40.7	39.4	46.9	46.5	46.4	45.6	0.0	0.0	-0.8
RO	35.0	36.6	36.4	36.0	35.9	n.a.	33.6	35.0	36.1	38.0	0.0	0.5	-1.6
PL	41.5	43.3	42.8	42.1	41.3	n.a.	41.2	41.5	42.6	42.6	0.0	0.7	0.2
SE	49.8	49.3	49.0	48.5	47.6	n.a.	49.4	49.9	49.8	49.5	-0.1	-0.5	-0.5
UK	38.2	38.3	38.0	37.8	37.7	37.7	41.1	40.6	40.9	40.9	-2.4	-2.6	-2.9
EU!	45.2	45.3	45.0	44.6	44.4	n.a.	45.8	45.6	45.8	45.7	-0.4	-0.5	-0.7

Note: EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

10010 7 (1.0.	0110010	rai balan			/								
											Difference	compar	ed to
1	201	and Conve	grammes	Commis	sion 2019 s	spring fored	forecast (red means higher in						
											programme)		
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	-1.0	-0.9	-0.3	-0.1	-0.1	n.a.	-1.4	-1.4	-1.4	-1.8	0.4	0.5	1.4
CY	2.3	1.5	1.0	0.8	0.6	n.a.	1.3	2.0	1.1	0.7	0.3	0.4	0.3
DE	1.6	0.8	0.6	0.5	0.4	0.6	0.8	1.6	1.1	0.8	0.0	-0.3	-0.2
EE	-1.9	-1.5	-1.6	-1.8	-2.0	-1.5	-1.7	-2.2	-1.7	-1.5	0.2	0.2	0.0
EL	4.8	3.1	2.0	1.5	1.3	n.a.	5.1	5.0	1.9	0.8	-0.2	1.1	1.2
IE	-1.6	-1.0	-0.5	0.4	1.1	1.7	-0.9	-1.4	-1.2	-0.5	-0.2	0.2	0.1
ES	-2.6	-2.7	-2.1	-1.6	-1.3	n.a.	-2.7	-2.7	-2.9	-3.2	0.1	0.2	1.2
FR	-2.5	-2.3	-2.2	-2.0	-1.7	n.a.	-2.7	-2.6	-2.6	-2.5	0.0	0.2	0.3
IT	-2.1	-2.3	-2.1	-1.8	-1.6	n.a.	-2.1	-2.2	-2.4	-3.6	0.1	0.1	1.5
LV	-2.1	-1.4	-0.8	-0.3	-0.2	n.a.	-1.2	-2.1	-1.6	-1.1	0.0	0.3	0.3
LT	-0.8	-0.8	-0.4	-0.1	0.1	n.a.	-0.8	-0.8	-1.0	-0.9	0.0	0.2	0.5
LU	2.3	0.6	0.6	0.6	1.2	1.5	1.5	2.1	0.9	0.5	0.1	-0.3	0.1
MT	1.5	0.6	0.9	1.1	1.1	n.a.	3.1	1.4	0.6	0.7	0.1	0.0	0.2
NL	0.8	0.4	0.5	-0.1	-0.2	n.a.	0.6	0.8	0.7	0.2	0.0	-0.3	0.2
AT	-0.4	-0.1	-0.1	0.1	0.0	0.1	-0.8	-0.5	-0.1	0.0	0.0	0.0	-0.1
PT	-0.2	-0.1	0.3	0.4	0.3	0.2	-1.3	-0.4	-0.5	-0.5	0.2	0.4	0.8
SI	-0.7	-0.6	-0.3	0.2	0.8	n.a.	-0.5	-0.7	-0.8	-0.3	0.0	0.2	-0.1
SK	-1.1	-0.7	-0.7	-0.6	-0.3	n.a.	-0.9	-1.3	-1.3	-1.4	0.1	0.6	0.7
FI	-0.8	-0.7	-0.3	-0.4	-0.6	n.a.	-0.7	-1.0	-1.0	-0.6	0.2	0.3	0.3
EA	-0.6	-0.9	-0.8	-0.7	-0.6	n.a.	-0.9	-0.7	-0.9	-1.2	0.0	0.0	0.5
BG	1.9	-0.4	0.3	0.0	-0.1	n.a.	1.1	1.9	0.7	0.6	0.0	-1.0	-0.3
CZ	0.5	0.0	-0.3	-0.4	-0.5	n.a.	1.0	0.4	-0.1	-0.4	0.1	0.2	0.0
DK	0.9	0.9	1.1	1.0	0.8	n.a.	1.5	0.8	0.9	1.0	0.1	0.1	0.1
HR	0.2	-0.8	-0.5	-0.4	-0.1	n.a.	0.6	-0.3	-0.8	-0.5	0.6	0.0	0.0
HU	-3.5	-3.0	-2.5	-1.9	-1.1	-0.4	-3.4	-3.7	-3.3	-2.7	0.2	0.3	0.3
RO	-2.6	-2.7	-2.9	-2.6	-2.3	n.a.	-2.9	-3.0	-3.6	-4.8	0.4	0.9	1.9
PL	-1.5	-2.8	-1.8	-1.1	-1.1	n.a.	-1.9	-1.4	-2.8	-3.0	-0.1	0.0	1.2
SE	0.5	0.6	0.9	1.4	2.1	n.a.	1.2	0.6	0.5	0.6	-0.2	0.2	0.3
UK	-1.7	-1.6	-1.1	-1.1	-0.7	-0.6	-2.7	-1.7	-1.7	-1.3	0.1	0.1	0.2
EU	-0.8	-1.0	-0.8	-0.7	-0.5	n.a.	-1.1	-0.8	-1.1	-1.2	0.0	0.1	0.4

Table A1.6: Structural balance (% of pot. GDP)

Note: For SCPs: recalculated by the European Commission on the basis of the information in the programme according to the commonly-agreed methodology. EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs
	201	9 Stability	and Conve	rgence Pro	grammes	. ,	Commiss	ion 2019 s	pring forec	Difference compared forecast (red means higher programme)			
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	0.4	0.1	0.6	0.3	0.0	n.a.	0.8	0.0	0.0	-0.3	0.3	0.2	0.9
CY	1.6	-0.8	-0.5	-0.1	-0.3	n.a.	0.2	0.7	-0.9	-0.4	0.9	0.1	0.0
DE	0.5	-0.7	-0.2	-0.2	-0.1	0.2	0.1	0.7	-0.5	-0.2	-0.2	-0.3	0.0
EE	0.0	0.4	-0.1	-0.2	-0.3	0.5	-0.9	-0.5	0.5	0.1	0.5	-0.1	-0.2
EL	1.0	-1.7	-1.0	-0.5	-0.2	n.a.	-0.5	-0.1	-3.1	-1.2	1.1	1.4	0.1
IE	0.2	0.5	0.5	0.9	0.7	n.a.	1.1	-0.4	0.2	0.7	0.6	0.4	-0.1
ES	0.8	-0.1	0.7	0.5	0.3	n.a.	0.4	0.0	-0.2	-0.3	0.8	0.1	1.0
FR	0.4	0.2	0.1	0.2	0.3	n.a.	0.1	0.2	0.0	0.0	0.3	0.2	0.1
IT	0.2	-0.2	0.2	0.2	0.2	n.a.	-0.4	-0.1	-0.2	-1.2	0.2	0.0	1.4
LV	-0.4	0.7	0.6	0.5	0.0	n.a.	-1.0	-0.8	0.4	0.6	0.4	0.3	0.1
LT	0.3	0.0	0.4	0.2	0.2	n.a.	-0.4	0.0	-0.2	0.1	0.3	0.2	0.3
LU	1.1	-1.7	0.0	0.0	0.5	0.3	-0.2	0.6	-1.2	-0.4	0.5	-0.5	0.4
MT	-1.8	-0.9	0.4	0.2	0.0	n.a.	2.9	-1.8	-0.8	0.1	0.0	-0.2	0.2
NL	0.6	-0.4	0.1	-0.5	-0.1	n.a.	0.3	0.2	-0.1	-0.5	0.5	-0.3	0.6
AT	0.9	0.3	0.0	0.2	-0.1	0.1	0.3	0.3	0.4	0.1	0.5	0.0	-0.1
PT	1.2	0.1	0.3	0.1	0.0	-0.1	0.7	0.9	-0.1	0.0	0.3	0.2	0.3
SI	0.7	0.2	0.2	0.6	0.6	n.a.	0.5	-0.2	-0.1	0.5	0.9	0.2	-0.3
SK	0.1	0.5	0.0	0.1	0.3	n.a.	1.1	-0.3	0.0	-0.2	0.4	0.5	0.1
FI	0.3	0.1	0.5	-0.1	-0.2	n.a.	0.1	-0.3	0.0	0.4	0.6	0.1	0.1
EA	0.5	-0.2	0.1	0.1	0.1	n.a.	0.1	0.3	-0.2	-0.3	0.2	0.0	0.4
BG	0.8	-2.3	0.7	-0.2	-0.2	n.a.	0.8	0.8	-1.3	0.0	0.0	-1.0	0.7
CZ	-0.6	-0.5	-0.3	-0.1	-0.1	n.a.	0.2	-0.6	-0.5	-0.2	0.0	0.1	-0.1
DK	-0.9	0.0	0.1	-0.1	-0.1	n.a.	1.2	-0.7	0.1	0.1	-0.2	0.0	0.0
HR	0.0	-1.1	0.3	0.1	0.3	n.a.	1.2	-1.0	-0.5	0.3	1.0	-0.6	0.1
HU	0.4	0.6	0.5	0.5	0.8	0.7	-1.6	-0.3	0.4	0.6	0.7	0.2	-0.1
RO	0.0	-0.2	-0.2	0.3	0.3	n.a.	-1.2	0.0	-0.7	-1.2	0.0	0.5	1.0
PL	1.1	-1.3	1.0	0.7	0.0	n.a.	0.0	0.5	-1.4	-0.2	0.6	0.1	1.2
SE	-0.7	0.1	0.3	0.5	0.6	n.a.	0.5	-0.6	-0.2	0.1	-0.1	0.3	0.2
UK	0.7	0.1	0.5	0.0	0.4	0.2	0.1	0.9	0.0	0.4	-0.3	0.1	0.0
EI I	0.5	-0.2	0.2	0.1	0.2	na	0.1	0.3	0.2	0.2	0.2	0.0	0.4

Table A1 7· al bala Ch in stri ctu of r of GDP) 1.

Note: For SCPs: recalculated by the European Commission on the basis of the information in the programme according to the commonly-agreed methodology. EU averages are based on UK Fiscal year data. **Source:** European Commission based on Commission 2019 spring forecast and 201 SCPs

Table A1.8:	Structur	al prima	y balanc	:e (% of p	oot. GDP)									
				_							Difference	compar	ed to	
	2019 Stability and Convergence Programmes							sion 2019 s	spring forec	ast	forecast (red means higher in			
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020	
BE	1.3	1.2	1.6	1.7	1.7	n.a.	1.0	0.9	0.7	0.3	0.4	0.5	1.3	
CY	4.8	3.8	3.2	2.9	2.5	n.a.	3.9	4.5	3.5	2.7	0.3	0.3	0.4	
DE	2.5	1.7	1.4	1.2	1.2	1.4	1.8	2.5	2.0	1.7	0.0	-0.3	-0.3	
EE	-1.9	-1.5	-1.6	-1.7	-1.9	-1.4	-1.6	-2.1	-1.7	-1.5	0.2	0.1	-0.1	
EL	8.1	6.2	4.9	4.4	4.2	n.a.	8.2	8.3	5.5	4.5	-0.3	0.7	0.5	
IE	0.0	0.4	0.7	1.5	2.2	2.9	1.1	0.3	0.3	0.7	-0.3	0.1	0.0	
ES	-0.2	-0.4	0.2	0.6	0.9	n.a.	-0.1	-0.2	-0.7	-1.1	0.1	0.3	1.4	
FR	-0.8	-0.8	-0.7	-0.4	0.0	n.a.	-1.0	-0.9	-0.9	-0.9	0.0	0.1	0.2	
IT	1.6	1.3	1.5	1.9	2.3	n.a.	1.7	1.5	1.2	0.2	0.1	0.1	1.4	
LV	-1.4	-0.7	0.0	0.5	0.6	n.a.	-0.3	-1.4	-1.0	-0.4	0.0	0.3	0.5	
LT	0.1	0.1	0.3	0.4	0.4	n.a.	0.4	0.1	-0.1	-0.2	0.0	0.3	0.6	
LU	2.6	0.8	0.9	0.8	1.3	1.6	1.9	2.5	1.2	0.8	0.1	-0.3	0.1	
MT	3.0	1.9	2.2	2.3	2.2	n.a.	5.0	2.9	2.0	2.0	0.1	-0.1	0.2	
NL	1.7	1.1	1.1	0.5	0.4	n.a.	1.6	1.7	1.5	0.9	0.0	-0.4	0.2	
AT	1.3	1.4	1.3	1.3	1.1	1.2	1.0	1.2	1.4	1.4	0.0	0.0	-0.1	
PT	3.2	3.2	3.3	3.2	3.0	2.9	2.5	3.0	2.8	2.5	0.2	0.4	0.8	
SI	1.3	1.1	1.1	1.5	2.0	n.a.	2.0	1.2	0.8	1.2	0.0	0.2	-0.1	
SK	0.2	0.5	0.3	0.4	0.7	n.a.	0.4	0.0	0.0	-0.2	0.1	0.5	0.6	
FI	0.0	0.1	0.6	0.5	0.3	n.a.	0.3	-0.1	-0.1	0.2	0.2	0.3	0.3	
EA	1.2	0.8	0.9	1.0	1.1	n.a.	1.0	1.2	0.8	0.5	0.0	0.0	0.4	
BG	2.6	0.2	0.9	0.5	0.4	n.a.	1.9	2.6	1.2	1.2	0.0	-1.0	-0.3	
CZ	1.2	0.8	0.5	0.3	0.2	n.a.	1.8	1.2	0.6	0.4	0.1	0.1	0.0	
DK	1.9	1.9	1.9	1.8	1.7	n.a.	2.6	1.9	2.0	2.1	0.0	-0.1	-0.1	
нк	2.5	1.5	1.5	1.5	1.6	n.a.	3.3	2.0	1.3	1.3	0.6	0.1	0.2	
HU	-1.0	-0.5	-0.1	0.3	1.1	1.7	-0.6	-1.2	-0.9	-0.4	0.2	0.4	0.3	
RU	-1.4	-1.5	-1.6	-1.4	-1.2	n.a.	-1.7	-1.8	-2.4	-3.5	0.4	0.9	1.9	
PL	-0.1	-1.4	-0.4	0.2	0.1	n.a.	-0.4	0.0	-1.4	-1.7	-0.1	0.0	1.3	
	1.0	1.0	1.3	1.7	2.4	n.a. 1 4	1.7	1.1	1.0	1.1	-0.2	0.0	0.2	
UK	0.6	0.8	1.1	1.0	1.4	1.4	-0.3	0.0	0.6	0.6	0.6	0.2	0.5	
20	1.0	0.7	0.9	0.9	1.1	II.d.	0.8	0.9	0.7	0.4	U. I	0.0	0.4	

Note: For SCPs: recalculated by the European Commission on the basis of the information in the programme according to the commonly-agreed methodology. EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

TUDIE AT.7.	Change	e în struct	Indi buin	lary baiu	nce (pps	. or poi	. GDF)						
	201	9 Stability	and Conve	rgence Pro	grammes		Commiss	sion 2019 s	pring forec	Difference compared to forecast (red means higher in programme)			
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	0.2	-0.1	0.4	0.2	0.0	n.a.	0.5	-0.2	-0.2	-0.4	0.4	0.1	0.8
CY	1.5	-1.0	-0.6	-0.3	-0.4	n.a.	0.0	0.6	-1.0	-0.8	0.9	0.0	0.2
DE	0.4	-0.8	-0.3	-0.2	-0.1	0.2	0.0	0.6	-0.5	-0.3	-0.2	-0.3	0.0
EE	-0.1	0.4	-0.1	-0.1	-0.3	0.5	-0.9	-0.5	0.5	0.1	0.4	-0.1	-0.2
EL EL	1.2	-1.9	-1.2	-0.5	-0.2	n.a.	-0.5	0.1	-2.9	-1.0	/ 1.0	1.0	-0.2
IE	-0.2	0.3	0.3	0.8	0.7	n.a.	0.9	-0.8	0.0	0.4	0.6	0.4	-0.1
ES	0.7	-0.2	0.6	0.4	0.3	n.a.	0.1	-0.1	-0.4	-0.5	0.8	0.2	1.1
FR	0.4	0.0	0.1	0.3	0.4	n.a.	0.0	0.1	-0.1	0.0	0.3	0.0	0.1
IT	0.1	-0.3	0.2	0.3	0.4	n.a.	-0.5	-0.2	-0.3	-1.1	0.2	0.0	1.3
LV	-0.6	0.7	0.7	0.5	0.0	n.a.	-1.1	-1.1	0.4	0.5	0.4	0.3	0.2
LT	0.0	0.0	0.2	0.0	0.0	n.a.	-0.6	-0.2	-0.2	-0.1	0.3	0.2	0.3
LU	1.1	-1.8	0.0	-0.1	0.5	0.3	-0.2	0.6	-1.3	-0.4	0.5	-0.5	0.4
MT	-2.0	-1.1	0.2	0.1	0.0	n.a.	2.6	-2.1	-0.9	0.0	0.0	-0.2	0.2
NL	0.5	-0.6	0.0	-0.5	-0.1	n.a.	0.1	0.1	-0.2	-0.6	0.5	-0.4	0.6
AT	0.7	0.1	-0.1	0.1	-0.2	0.0	0.0	0.2	0.2	0.0	0.5	0.0	-0.1
PT	0.8	0.0	0.1	-0.1	-0.2	-0.1	0.4	0.5	-0.3	-0.2	0.3	0.2	0.3
SI	0.2	-0.2	0.1	0.4	0.5	n.a.	0.0	-0.7	-0.4	0.4	.0.9	0.2	-0.3
SK	0.0	0.3	-0.1	0.1	0.3	n.a.	0.9	-0.4	-0.1	-0.2	0.4	0.4	0.1
FI	0.2	0.1	0.4	-0.1	-0.2	n.a.	0.0	-0.4	0.0	0.4	÷ 0.6	0.1	0.1
EA	0.4	-0.4	0.1	0.1	0.1	n.a.	0.0	0.1	-0.3	-0.3	0.2	0.0	0.4
BG	0.7	-2.4	0.7	-0.3	-0.2	n.a.	0.7	0.7	-1.4	-0.1	0.1	-1.1	0.8
CZ	-0.6	-0.5	-0.3	-0.1	-0.1	n.a.	0.0	-0.6	-0.5	-0.2	. 0.0	0.1	-0.1
DK	-0.9	-0.1	0.1	-0.1	-0.1	n.a.	0.9	-0.7	0.1	0.1	-0.2	-0.1	0.0
HR	-0.3	-1.1	0.0	0.0	0.1	n.a.	0.8	-1.3	-0.6	-0.1	1.0	-0.4	0.1
HU	0.1	0.6	0.4	0.3	0.8	0.6	-2.0	-0.6	0.3	0.5	0.7	0.3	-0.1
RO	-0.1	-0.2	-0.1	0.2	0.2	n.a.	-1.4	-0.1	-0.6	-1.1	0.0	0.4	1.0
PL	1.0	-1.3	1.0	0.6	-0.1	n.a.	-0.2	0.4	-1.4	-0.3	0.6	0.2	1.3
SE	-0.7	0.0	0.3	0.4	0.6	n.a.	0.5	-0.6	-0.2	0.2	-0.1	0.2	0.2
UK	0.3	0.2	0.3	-0.1	0.4	0.1	0.5	0.3	0.6	0.0	0.0	-0.4	0.3
EU	0.3	-0.3	0.1	0.1	0.2	n.a.	0.1	0.1	-0.2	-0.3	0.2 از	-0.1	0.4

Table A1.9: Change in structural primary balance (pps of not GDP)

Note: For SCPs: recalculated by the European Commission based on the information in the programme according to the commonly-agreed methodology. EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

TUDIE AT.T	0. FISCULE	anon pus	eu on me	expend	liule bell	chinan)					
	201	9 Stability	and Conve	rgence Pro	grammes		Commis	sion 2019 s	spring forec	Difference compared to forecast (red means higher in programme)			
	2018	2019	2020	2021	2022	2023	2017	2018	2019	2020	2018	2019	2020
BE	n.a.	0.4	0.5	0.2	-0.2	n.a.	0.4	-0.4	-0.5	-0.6	n.a.	0.9	1.1
CY	n.a.	-3.5	-0.2	-0.2	0.4	n.a.	-1.3	2.0	-4.0	-0.5	n.a.	0.5	0.2
DE	n.a.	-0.8	-0.2	-0.3	-0.1	n.a.	-0.3	-0.2	-0.7	-0.4	n.a.	0.0	0.2
EE	n.a.	0.5	0.1	0.4	0.1	n.a.	-0.1	-0.2	-0.1	-0.1	n.a.	0.6	0.2
EL	n.a.	-1.6	0.4	0.3	0.0	n.a.	1.2	-0.8	-0.7	-0.5	n.a.	-0.9	0.9
IE	n.a.	0.8	0.8	0.8	0.6	n.a.	0.2	0.5	0.5	1.2	n.a.	0.2	-0.4
ES	n.a.	-0.4	-0.1	-0.1	-0.2	n.a.	0.0	-1.0	-0.6	-0.6	n.a.	0.2	0.5
FR	n.a.	-0.1	-0.1	0.1	0.3	n.a.	-0.2	0.1	-0.2	-0.2	n.a.	0.1	0.1
IT	n.a.	-0.4	0.3	0.1	0.3	n.a.	-0.4	-0.5	-0.4	-1.0	n.a.	0.0	1.3
LV	n.a.	0.9	-0.8	0.7	0.3	n.a.	-0.6	-1.3	-0.2	-0.3	n.a.	1.1	-0.5
LT	n.a.	-0.7	0.1	-0.4	-0.8	n.a.	0.3	-0.8	-0.8	0.0	n.a.	0.1	0.1
LU	n.a.	-0.4	-0.4	-0.3	-0.1	n.a.	-2.3	-0.7	-1.2	-0.7	n.a.	0.8	0.3
MT	n.a.	-1.0	0.3	0.6	0.5	n.a.	0.8	-2.1	-0.7	0.2	n.a.	-0.2	0.1
NL	n.a.	-0.7	-0.1	-0.2	0.2	n.a.	0.3	-0.6	-0.5	-0.6	i n.a.	-0.2	0.5
AT	n.a.	0.0	0.0	0.0	-0.1	n.a.	0.0	-0.4	0.0	0.0	n.a.	0.0	0.0
PT	n.a.	-0.6	-0.2	-0.2	-0.2	n.a.	0.1	-0.8	-1.0	-0.4	n.a.	0.4	0.2
SI	n.a.	-0.3	0.5	0.0	0.2	n.a.	0.4	-0.2	-0.9	0.3	n.a.	0.6	0.3
SK	n.a.	0.6	0.4	0.3	0.8	n.a.	0.8	-0.5	0.0	0.0	n.a.	0.5	0.4
FI	n.a.	-0.6	0.3	-0.1	0.1	n.a.	0.0	-0.2	-0.5	0.2	n.a.	-0.1	0.0
EA	n.a.	-0.3	0.0	0.0	0.1	n.a.	-0.1	-0.3	-0.5	-0.4	n.a.	0.2	0.4
BG	n.a.	-1.2	0.3	0.1	0.1	n.a.	-0.3	1.3	-0.5	0.2	n.a.	-0.7	0.2
CZ	n.a.	-0.6	-0.3	-0.1	0.1	n.a.	-0.1	-1.3	-0.9	-0.3	n.a.	0.4	0.1
DK	n.a.	1.2	0.3	0.0	-0.2	n.a.	0.3	-0.3	0.3	0.3	n.a.	0.9	0.1
HR	n.a.	-1.7	-0.7	-0.3	0.1	n.a.	0.2	-1.2	-1.5	-0.3	n.a.	-0.2	-0.4
HU	n.a.	0.1	-0.5	-0.1	0.0	n.a.	-2.3	0.3	0.2	-0.4	n.a.	-0.1	0.0
RO	n.a.	-0.7	-0.1	-0.3	-0.7	n.a.	-2.5	-0.6	-0.7	-1.2	n.a.	0.0	1.1
PL	n.a.	-1.8	-0.4	1.0	0.7	n.a.	0.3	0.3	-1.3	-0.8	n.a.	-0.5	0.4
SE	n.a.	0.1	0.5	0.4	0.6	n.a.	0.6	-0.5	0.3	0.4	n.a.	-0.2	0.0
UK	n.a.	0.2	0.2	0.0	-0.2	n.a.	0.1	0.2	0.1	0.1	n.a.	0.1	0.1
EU	n.a.	-0.3	0.0	0.0	0.1	n.a.	0.4	-0.2	-0.6	-0.4	n.a.	0.3	0.4

Table A1.10: Fiscal effort based on the expenditure benchmark (% of GDP)

Note: Calculated based on the 10-year average potential growth rate as estimated by the Commission 2019 spring forecast. EU averages are based on UK Fiscal year data. Source: European Commission based on Commission 2019 spring forecast and 2019 SCPs

2018 2019 2020 2021 2022 BE 102.0 100.6 98.5 96.2 94.0 CY 102.5 95.7 89.1 83.0 77.5 DE 61 58 3/4 56 2/4 54 3/4 53 EE 7.9 8.2 8.1 8.0 8.3 EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2		Commis	sion 2019	spring fore	Difference forecast (red programme)	compar means I	ed to nigher in	
BE 102.0 100.6 98.5 96.2 94.0 CY 102.5 95.7 89.1 83.0 77.5 DE 61 583/4 562/4 543/4 53 EE 7.9 8.2 8.1 8.0 8.3 EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1	2023	2017	2018	2019	2020	2018	2019	2020
CY 102.5 95.7 89.1 83.0 77.5 DE 61 58.3/4 56.2/4 54.3/4 53 EE 7.9 8.2 8.1 8.0 8.3 EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5	n.a.	103.4	102.0	101.3	100.7	0.0	-0.7	-2.2
DE 61 58 3/4 56 2/4 54 3/4 53 EE 7.9 8.2 8.1 8.0 8.3 EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	n.a.	95.8	102.5	96.4	89.9	0.0	-0.8	-0.8
EE 7.9 8.2 8.1 8.0 8.3 EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	51 1/3	64.5	60.9	58.4	55.6	0.0	0.2	1.0
EL 181.1 170.6 163.9 157.5 153.3 IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	8.2	9.2	8.4	8.5	8.5	-0.5	-0.3	-0.4
IE 64.8 61.1 55.8 55.4 53.2 ES 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 10.4 41.6 42.7 40.0 42.7	n.a.	176.2	181.1	174.9	168.9	0.0	-4.3	-5.0
FS 97.1 95.8 94.0 91.4 88.7 FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	n.a.	68.5	64.8	61.3	55.9	0.0	-0.2	-0.1
FR 98.4 98.9 98.7 98.1 96.8 IT 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	n.a.	98.1	97.1	96.3	95.7	0.0	-0.4	-1.7
II 132.2 132.6 131.3 130.2 128.9 LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 104.5 440.1 47.1 45.7 44.6	n.a.	98.4	98.4	99.0	98.9	0.0	-0.1	-0.2
LV 35.9 37.4 36.1 33.5 33.1 LT 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 4245 440.0 42.7 40.0 40.0 7	n.a.	131.4	132.2	133.7	135.2	0.0	-1.1	-3.9
LI 34.2 37.0 36.2 35.4 32.9 LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 424.5 440.6 445.2 400.0 400.7	n.a.	40.0	35.9	34.5	33.5	0.0	2.9	2.6
LU 21.4 20.2 19.9 19.3 18.4 MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 104.5 140.2 102.7 100.7	n.a.	39.4	34.2	37.0	36.4	0.0	0.0	-0.2
MT 46.0 42.7 39.4 36.2 33.2 NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8 DT 4245 449.0 445.2 400.0 402.7	17.3	23.0	21.4	20.7	20.3	0.0	-0.5	-0.4
NL 52.4 49.1 47.1 45.7 44.6 AT 73.8 69.6 66.5 64.0 61.8	n.a.	50.2	46.0	42.8	40.2	0.0	-0.1	-0.8
AI 73.8 69.6 66.5 64.0 61.8	n.a.	57.0	52.4	49.1	46.7	0.0	0.0	0.4
	59.8	78.2	73.8	69.7	66.8	0.0	-0.1	-0.2
P1 121.5 118.6 115.2 109.0 103.7	99.6	124.8	121.5	119.5	116.6	0.0	-0.9	-1.4
SI 70.1 65.4 61.3 57.9 54.7	n.a.	74.1	70.1	65.9	61.7	0.0	-0.5	-0.4
SK 48.9 47.5 45.9 44.9 44.4	n.a.	50.9	48.9	47.3	46.0	0.0	0.2	-0.1
FI 58.9 58.1 57.4 57.4 57.7	n.a.	61.3	58.9	58.3	57.7	0.0	-0.2	-0.3
EA 87.1 85.5 83.7 82.0 80.2	n.a.	89.1	87.1	85.8	84.3	0.0	-0.3	-0.6
BG 22.0 20.0 19.1 17.8 10.7	n.a.	25.0	22.0	20.5	18.4	0.0	0.1	0.7
0Z 32.7 31.3 30.8 30.2 29.7	n.a.	34.7	32.7	31.7	31.1	0.0	-0.2	-0.3
DK 33.8 33.4 33.4 34.0 35.5	n.a.	35.5	34.1	33.0	32.5	-0.3	0.4	0.8
	n.a.	77.8	74.0	70.9	67.0	0.0	0.7	0.9
PO 250 254 254 250 240	55.9	73.4	70.8	09.2	207.7	0.0	0.0	-1.0
RU 30.0 30.4 30.4 35.2 34.8	n.a.	35.2	35.0	30.0	38.4	0.0	-0.0	-3.0
FL 40.9 47.9 40.0 42.9 40.0	n.a.	50.6	48.9	48.2	47.4	0.0	-0.3	-1.4
	n.a.	40.8	38.8	34.4	32.4	0.0	0.1	0.4
	80.0	80.4	80.4	80.0	02.1 79.6	0.1	0.0	0.2

Table A1.11: General government total debt (% of GDP)

Note: in case of missing programmes: weighted average of the figures for those countries that have submitted a programme. Commission EU averages are based on UK Fiscal year data. **Source:** European Commission based on Commission 2019 spring forecast and 2019 SCPs

EUROPEAN ECONOMY INSTITUTIONAL SERIES

European Economy Institutional series can be accessed and downloaded free of charge from the following address:

https://ec.europa.eu/info/publications/economic-and-financial-affairspublications en?field eurovoc taxonomy target id selective=All&field core nal countries tid selective=All &field core date published value[vear]=All&field core tags tid i18n=22621.

Titles published before July 2015 can be accessed and downloaded free of charge from:

- <u>http://ec.europa.eu/economy_finance/publications/european_economy/index_en.htm</u> (the main reports, e.g. Economic Forecasts)
- <u>http://ec.europa.eu/economy_finance/publications/occasional_paper/index_en.htm</u> (the Occasional Papers)
- <u>http://ec.europa.eu/economy_finance/publications/qr_euro_area/index_en.htm</u> (the Quarterly Reports on the Euro Area)

GETTING IN TOUCH WITH THE EU

In person

All over the European Union there are hundreds of Europe Direct Information Centres. You can find the address of the centre nearest you at: <u>http://europa.eu/contact</u>.

On the phone or by e-mail

Europe Direct is a service that answers your questions about the European Union. You can contact this service:

- by freephone: 00 800 6 7 8 9 10 11 (certain operators may charge for these calls),
- at the following standard number: +32 22999696 or
- by electronic mail via: <u>http://europa.eu/contact.</u>

FINDING INFORMATION ABOUT THE EU

Online

Information about the European Union in all the official languages of the EU is available on the Europa website at: <u>http://europa.eu</u>.

EU Publications

You can download or order free and priced EU publications from EU Bookshop at: <u>http://publications.europa.eu/bookshop</u>. Multiple copies of free publications may be obtained by contacting Europe Direct or your local information centre (see <u>http://europa.eu/contact</u>).

EU law and related documents

For access to legal information from the EU, including all EU law since 1951 in all the official language versions, go to EUR-Lex at: <u>http://eur-lex.europa.eu</u>.

Open data from the EU

The EU Open Data Portal (<u>http://data.europa.eu/euodp/en/data</u>) provides access to datasets from the EU. Data can be downloaded and reused for free, both for commercial and non-commercial purposes.

